PATENT ASSIGNMENT

Electronic Version v1.1 Stylesheet Version v1.1

SUBMISSION TYPE:	NEW ASSIGNMENT
NATURE OF CONVEYANCE:	ASSIGNMENT

CONVEYING PARTY DATA

Name	Execution Date
CORE LABORATORIES CANADA LIMITED	04/22/2004
CORE LABORATORIES LP	04/22/2004
CORE LAB DE MEXICO S.A. DE C.V.	04/22/2004
CORE LABORATORIES N.V.	04/22/2004

RECEIVING PARTY DATA

Name:	PARADIGM GEOTECHNOLOGY B.V.	
Street Address:	A-TOWER, 7TH FLOOR	
Internal Address:	STRAWINSKYLAAN 717	
City:	AMSTERDAM	
State/Country:	NETHERLANDS	
Postal Code:	1077 XX	

PROPERTY NUMBERS Total: 1

Property Type	Number
Patent Number:	6055482

CORRESPONDENCE DATA

 Fax Number:
 (646)878-0801

 Phone:
 646-878-0800

 Email:
 USPTO@pczlaw.com

Correspondence will be sent to the e-mail address first; if that is unsuccessful, it will be sent

via US Mail.

Correspondent Name: Pearl Cohen Zedek Latzer, LLP

Address Line 1: 1500 Broadway
Address Line 2: 12th Floor

Address Line 4: New York, NEW YORK 10036

ATTORNEY DOCKET NUMBER:	P-70472-US	
NAME OF SUBMITTER:	SHARON BRAUDE	PATENT

501995152 REEL: 028590 FRAME: 0001

6055482

00 07, HO

Total Attachments: 104
source=P-70472-US-FASS-19JUL12#page1.tif
source=P-70472-US-FASS-19JUL12#page2.tif
source=P-70472-US-FASS-19JUL12#page3.tif
source=P-70472-US-FASS-19JUL12#page4.tif
source=P-70472-US-FASS-19JUL12#page5.tif
source=P-70472-US-FASS-19JUL12#page6.tif
source=P-70472-US-FASS-19JUL12#page7.tif
source=P-70472-US-FASS-19JUL12#page8.tif
source=P-70472-US-FASS-19JUL12#page9.tif
source=P-70472-US-FASS-19JUL12#page10.tif
source=P-70472-US-FASS-19JUL12#page11.tif
source=P-70472-US-FASS-19JUL12#page12.tif
source=P-70472-US-FASS-19JUL12#page13.tif
source=P-70472-US-FASS-19JUL12#page14.tif
source=P-70472-US-FASS-19JUL12#page15.tif
source=P-70472-US-FASS-19JUL12#page16.tif
source=P-70472-US-FASS-19JUL12#page17.tif
source=P-70472-US-FASS-19JUL12#page18.tif
source=P-70472-US-FASS-19JUL12#page19.tif
source=P-70472-US-FASS-19JUL12#page20.tif
source=P-70472-US-FASS-19JUL12#page21.tif
source=P-70472-US-FASS-19JUL12#page22.tif
source=P-70472-US-FASS-19JUL12#page23.tif
source=P-70472-US-FASS-19JUL12#page24.tif
source=P-70472-US-FASS-19JUL12#page25.tif
source=P-70472-03-1A33-1930L12#page25.tif
source=P-70472-US-FASS-1930L12#page27.tif
source=P-70472-US-FASS-1930L12#page28.tif
source=P-70472-US-FASS-19JUL12#page29.tif
source=P-70472-03-FASS-1930L12#page29.tif
source=P-70472-US-FASS-19JUL12#page31.tif
source=P-70472-US-FASS-1930L12#page31.tif
source=P-70472-US-FASS-19JUL12#page33.tif
source=P-70472-US-FASS-19JUL12#page34.tif
source=P-70472-US-FASS-19JUL12#page35.tif
source=P-70472-US-FASS-19JUL12#page36.tif
source=P-70472-US-FASS-19JUL12#page37.tif
source=P-70472-US-FASS-19JUL12#page38.tif
source=P-70472-US-FASS-19JUL12#page39.tif
source=P-70472-US-FASS-19JUL12#page40.tif
source=P-70472-US-FASS-19JUL12#page41.tif
source=P-70472-US-FASS-19JUL12#page42.tif
source=P-70472-US-FASS-19JUL12#page43.tif
source=P-70472-US-FASS-19JUL12#page44.tif
source=P-70472-US-FASS-19JUL12#page45.tif
source=P-70472-US-FASS-19JUL12#page46.tif
source=P-70472-US-FASS-19JUL12#page47.tif
source=P-70472-US-FASS-19JUL12#page48.tif
source=P-70472-US-FASS-19JUL12#page49.tif
source=P-70472-US-FASS-19JUL12#page50.tif
source=P-70472-US-FASS-19JUL12#page51.tif
source=P-70472-US-FASS-19JUL12#page52.tif
source=P-70472-US-FASS-19JUL12#page53.tif
source=P-70472-US-FASS-19JUL12#page54.tif
I

PATENT REEL: 028590 FRAME: 0002 source=P-70472-US-FASS-19JUL12#page55.tif source=P-70472-US-FASS-19JUL12#page56.tif source=P-70472-US-FASS-19JUL12#page57.tif source=P-70472-US-FASS-19JUL12#page58.tif source=P-70472-US-FASS-19JUL12#page59.tif source=P-70472-US-FASS-19JUL12#page60.tif source=P-70472-US-FASS-19JUL12#page61.tif source=P-70472-US-FASS-19JUL12#page62.tif source=P-70472-US-FASS-19JUL12#page63.tif source=P-70472-US-FASS-19JUL12#page64.tif source=P-70472-US-FASS-19JUL12#page65.tif source=P-70472-US-FASS-19JUL12#page66.tif source=P-70472-US-FASS-19JUL12#page67.tif source=P-70472-US-FASS-19JUL12#page68.tif source=P-70472-US-FASS-19JUL12#page69.tif source=P-70472-US-FASS-19JUL12#page70.tif source=P-70472-US-FASS-19JUL12#page71.tif source=P-70472-US-FASS-19JUL12#page72.tif source=P-70472-US-FASS-19JUL12#page73.tif source=P-70472-US-FASS-19JUL12#page74.tif source=P-70472-US-FASS-19JUL12#page75.tif source=P-70472-US-FASS-19JUL12#page76.tif source=P-70472-US-FASS-19JUL12#page77.tif source=P-70472-US-FASS-19JUL12#page78.tif source=P-70472-US-FASS-19JUL12#page79.tif source=P-70472-US-FASS-19JUL12#page80.tif source=P-70472-US-FASS-19JUL12#page81.tif source=P-70472-US-FASS-19JUL12#page82.tif source=P-70472-US-FASS-19JUL12#page83.tif source=P-70472-US-FASS-19JUL12#page84.tif source=P-70472-US-FASS-19JUL12#page85.tif source=P-70472-US-FASS-19JUL12#page86.tif source=P-70472-US-FASS-19JUL12#page87.tif source=P-70472-US-FASS-19JUL12#page88.tif source=P-70472-US-FASS-19JUL12#page89.tif source=P-70472-US-FASS-19JUL12#page90.tif source=P-70472-US-FASS-19JUL12#page91.tif source=P-70472-US-FASS-19JUL12#page92.tif source=P-70472-US-FASS-19JUL12#page93.tif source=P-70472-US-FASS-19JUL12#page94.tif source=P-70472-US-FASS-19JUL12#page95.tif source=P-70472-US-FASS-19JUL12#page96.tif source=P-70472-US-FASS-19JUL12#page97.tif source=P-70472-US-FASS-19JUL12#page98.tif source=P-70472-US-FASS-19JUL12#page99.tif source=P-70472-US-FASS-19JUL12#page100.tif source=P-70472-US-FASS-19JUL12#page101.tif source=P-70472-US-FASS-19JUL12#page102.tif source=P-70472-US-FASS-19JUL12#page103.tif source=P-70472-US-FASS-19JUL12#page104.tif

> PATENT REEL: 028590 FRAME: 0003

Execution Version

ASSET PURCHASE AGREEMENT

between

CORE LABORATORIES CANADA LIMITED,

CORE LABORATORIES LP and

CORE LAB de MEXICO S.A. de C.V.,

as "Sellers"

CORE LABORATORIES N.V.

as "Seller Parent"

and

PARADIGM GEOTECHNOLOGY B.V.,
PARADIGM GEOPHYSICAL CANADA LTD.

and

PARADIGM GEOPHYSICAL CORP.

as "Buyers"

April 22, 2004

PATENT REEL: 028590 FRAME: 0004

TABLE OF CONTENTS

Page

1. Def	finitions; Interpretation	1
1.1	Defined Terms	1
1.2	Interpretation	
2. Pui	rchase and Sale of the Purchased Assets	2
2.1	Basic Transaction	
2.2	Excluded Assets	
2.3	Purchase Price	
2.4	Purchase Price Adjustments	
2.5	Assumed Liabilities; Excluded Liabilities	
2.6	Allocation of Purchase Price	
2.7	The Closing	
2.8	Deliveries at the Closing	
2.9	Prorations	
2.10	Nonassignable Items	
	* . ^ *********************************	A J
	presentations and Warranties of Seller Parties	
3.1	Organization and Qualification of the Seller Parties	
3.2	Authorization of Transaction	
3.3	Noncontravention	12
3.4	Consents	12
3.5	Purchased Assets	
3.6	Companies Involved with the Business	13
3.7	Tax Matters	
3.8	Leased Real Property	14
3.9	Legal Compliance	14
3.10	Environmental Matters	14
3.11	Powers of Attorney	15
3.12	Litigation	15
3.13	The Mexican Subsidiaries	15
3.14	Employee Matters	16
3.15	Intellectual Property	18
3.16	Material Agreements	19
3.17	Governmental Permits	19
3.18	Material Adverse Change	19
3.19	Brokers' Fees	
3.20	Accuracy of Information.	20
4. Rei	presentations and Warranties Concerning the Buyer	20
4.1	Organization of the Buyer	
4.2	Authorization of Transaction	
4.3	Noncontravention	
4.4	Consents	
T 0 T	~~####################################	41

4.5	Brokers' Fees	
4.6	Litigation	21
4.7	[Intentionally deleted.]	21
4.8	Due Diligence Investigation and Other Acknowledgments	21
5. [I	Intentionally deleted.]	22
6. P	ost-Closing Covenants	22
6.1	General; Certain Assets	
6.2	Transition	23
6.3	Information	
6.4	Transfer Taxes	25
6.5	Employment Obligations	26
6.6	Employee Benefit Plans	
6.7	Non-Use of Core Lab Marks	28
6.8	Insurance Coverage	29
6.9	No Competition	29
6.10	Collection of Accounts Receivable	30
7. [I	Intentionally deleted.]	31
8. Sı	urvival; Indemnification	31
8.1	Survival of Representations, Warranties and Covenants	31
8.2	Indemnification Provisions for Benefit of the Buyers	31
8.3	Indemnification Provisions for Benefit of the Seller Parties	32
8.4	Matters Involving Third Parties	32
8.5	Limitations on Sellers' Indemnification Obligations	33
8.6	Other Indemnification Provisions	34
8.7	Exclusive Remedy	
8.8	Disclaimer of Other Representations and Warranties	35
9. D i	ispute Resolution	36
9.1	Negotiations By Senior Management	36
9.2	Binding Arbitration.	36
10. M	liscellaneous	38
10.1	Press Releases and Public Announcements	38
10.2	Cooperation on Tax Matters; Retention of Records	38
10.3	No Third-Party Beneficiaries	39
10.4	Entire Agreement	39
10.5	Succession and Assignment	39
10.6	Counterparts	39
10.7	Headings	39
10.8	Notices	39
10.9	Governing Law	41
10.10	0 Amendments and Waivers	41
10.11	1 Severability	41
10.12	2 Expenses	41

10.13	Annexes, Exhibits, Schedules and Disclosure Schedules	41
10.14	Specific Performance	41 1
10.15	Waiver of Consequential Damages	. 44 3 . 411
10.16	Guaranty	41
10.17	Electronic Signatures	. 4 .2 . 41

ATTACHMENTS

Annex 1		Defined Terms
Annex 2		Assignment and Assumption Agreement
Annex 3		License Agreement
Annex 4		Sublease Agreement
Exhibit 1(a)	www.ca.ca.ca	Buyers' Required Consents
Exhibit 1(b)		Sellers' Required Consents
Exhibit 2.1(a)		Accounts Receivable
Exhibit 2.1(b)		Work in Process
Exhibit 2.1(c)		Tangible Personal Property
Exhibit 2.1(d)		Leased Real Property
Exhibit 2.1(f)	***************************************	Intellectual Property
Exhibit 2.1(g)		Governmental Permits
Exhibit 2.1(h)		Licenses
Exhibit 2.1(i)		Assigned Agreements
Exhibit 2.1(k)		Other Purchased Assets
Exhibit 2.2	-	Certain Excluded Assets
Exhibit 2.4(a)		March 31, 2004 Net Asset Statement
Exhibit 2.5		Certain Assumed Liabilities
Exhibit 2.6		Purchase Price Allocation
Exhibit 6.1		Certain Assets in Mexico
Exhibit 6.10	**************************************	Specified Receivable

Sellers' Disclosure Schedule Section 3.1 Organia

Section 3.1		Organization and Qualification of the Seller Parties
Section 3.2		Authorization
Section 3.3		Noncontravention
Section 3.4	***************************************	Consents
Section 3.5		Purchased Assets
Section 3.6		Companies Involved with the Business
Section 3.7		Tax Matters
Section 3.8	*****	Leased Real Property
Section 3.9		Legal Compliance
Section 3.10	PRO	Environmental Matters
Section 3.11	_	Powers of Attorney
Section 3.12		Litigation
Section 3.13	****	The Mexican Subsidiaries
Section 3.14		Employee Matters
Section 3.15		Intellectual Property
Section 3.16		Material Agreements
Section 3.17		Governmental Permits
Section 3.18		Material Adverse Change
Section 3.19		Brokers' Fees
Section 3.20	_	Accuracy of Information

Buyers' Disclosure Schedule

ASSET PURCHASE AGREEMENT

v

ASSET PURCHASE AGREEMENT

This ASSET PURCHASE AGREEMENT (this "Agreement") among Core Laboratories Canada Limited, an Alberta, Canada, corporation ("Core Lab Canada"), Core Laboratories LP, a Delaware limited partnership ("Core Lab LP"), Core Lab de Mexico S.A. de C.V., a Mexican sociedad anonima de capital variable ("Core Lab de Mexico") (each of the foregoing entities being called a "Seller"), Core Laboratories N.V., a Netherlands public limited liability company ("Seller Parent" and, collectively with the Sellers, the "Seller Parties"), and Paradigm Geotechnology B.V., a Netherlands private limited liability company ("Paradigm Geotechnology"), Paradigm Geophysical Canada Ltd., an Alberta, Canada corporation ("Paradigm Canada"), and Paradigm Geophysical Corp., a Delaware corporation ("Paradigm US," with each of the foregoing entities other than the Seller Parties being called a "Buyer"), is made as of April 22, 2004. Each of the Seller Parties and the Buyers are herein referred to individually as a "Party" and collectively as the "Parties."

RECITALS

The Sellers, which are wholly owned direct or indirect subsidiaries of Seller Parent, operate directly, or indirectly through the Mexican Subsidiaries, a seismic data processing, time and depth imaging, seismic signal enhancement and Advanced Reservoir Geophysics business on a worldwide basis known as the Reservoir Technologies Division of Sellers (the "Business") and own assets (including the equity interests in the Mexican Subsidiaries) related to the Business. The Buyers desire to purchase from the Sellers, and the Sellers desire to sell to the Buyers, the property and assets primarily used in or necessary to the operation of the Business, which transaction will benefit Seller Parent as the direct or indirect owner of the Sellers.

AGREEMENT

Accordingly, in consideration of the premises and other good and valuable consideration, the Parties intending to be legally bound agree as follows:

1. DEFINITIONS; INTERPRETATION.

1.1 Defined Terms.

Capitalized terms used in this Agreement shall, unless expressly stated otherwise, have the meanings specified in <u>Annex 1</u>.

1.2 Interpretation.

In this Agreement, unless the context indicates otherwise:

- (a) Terms defined in the singular have the corresponding meanings in the plural, and terms defined in the plural have the corresponding meanings in the singular.
 - (b) Words importing any gender include the other gender.

- (c) Unless otherwise specified, any reference to a time of day means U.S. Central Standard Time or U.S. Central Daylight Time, whichever is prevailing at such time.
- (d) References to statutes, sections or regulations are to be construed as including all statutory or regulatory provisions consolidating, amending, replacing, succeeding or supplementing the statute, section or regulation referred to, and including rules and regulations promulgated thereunder.
- (e) References to "writing" include printing, typing, lithography, facsimile reproduction and other means of reproducing words in a tangible visible form.
- (f) The words "including," "includes" and "include" shall be deemed to be followed by the words "without limitation" or "but not limited to" or words of similar import.
- (g) References to sections (or subdivisions of sections), exhibits, annexes or schedules are to those of this Agreement unless otherwise indicated.
- (h) References to agreements and other contractual instruments shall be deemed to include all attachments thereto.
- (i) References to Persons include their respective successors and permitted assigns.
- (j) All accounting terms used herein and not expressly defined herein have the meanings given to them under GAAP.
- (k) Except where Mexican pesos or Canadian dollars are indicated, references to "dollars" or "\$" are to dollars of the United States of America. If any amount or limit originates in another currency, it will be converted to dollars at the prevailing exchange rate at the time.

2. PURCHASE AND SALE OF THE PURCHASED ASSETS.

- 2.1 Basic Transaction. On the Closing Date, concurrently with the execution and delivery of this Agreement, the Sellers shall (and Seller Parent shall cause the Sellers to) sell, convey, assign, deliver and transfer to the Buyers, and the Buyers, allocated among them as they may determine, shall (and Paradigm Geotechnology shall cause the other Buyers to) buy and take possession of, (1) all real, personal, tangible, intangible, contingent and other properties, rights and other assets of any kind owned or held by the Sellers that are used primarily in, or necessary to the operation of, the Non-Mexican Business, except the Excluded Assets and (2) all of the issued and outstanding capital shares of the Mexican Subsidiaries (collectively, the "Purchased Assets"). The Purchased Assets include the following assets owned or held by the Sellers:
 - (a) Receivables. All accounts, notes and other receivables arising from the Non-Mexican Business as of the Closing Date, other than accounts, notes and other

receivables from an Affiliate of any Seller ("Receivables"), including those Receivables listed on Exhibit 2.1(a).

- (b) *Inventories*. All work-in-process and supplies used or allocated to the Non-Mexican Business ("*Inventories*"), including the work-in-process listed on <u>Exhibit 2.1(b)</u>.
- (c) Personal Property. All other tangible personal property (excluding Inventories) owned, leased or used primarily in association with the Non-Mexican Business ("Tangible Personal Property"), including all equipment, machinery, tools, vehicles, computer equipment and related software, office furniture and fixtures described on Exhibit 2.1(c), together with all related leases, warranties, and other rights of use.
- (d) Leased Real Property. All the leases of real property held specifically in connection with the operations of the Non-Mexican Business ("Leased Real Property"), including leasehold interests specifically associated with the Business set forth on Exhibit 2.1(d).
- (e) Records. Copies of assets ledger, inventory record, budget, customer lists, customer credit information, supplier lists, technical data, sales literature, books, corporate minute books of the Mexican Subsidiaries (but excluding all other corporate minute books), notes, files and all other operating records and other graphic or electronically stored operating information specifically associated or used in connection with the Non-Mexican Business.
- (f) Intellectual Property. All trademarks, trade names, service marks, copyrights, patents, patent applications, inventions, formulas, trade secrets, know-how, designs, processes and similar intangibles specifically associated or used in connection with the Business ("Intellectual Property"), including those items listed on Exhibit 2.1(f), but specifically excluding (i) subject to the License Agreement from Core Laboratories Global N.V. ("Core Lab Global") of the technology represented by the Coherence Cube Patents, ownership of the Coherence Cube Patents and (ii) any interest in any lawsuits pending as of the Closing Date (and any lawsuit to be brought after the Closing Date with Buyers' consent, not to be unreasonably withheld) and any licenses or settlements based upon the Coherence Cube Patents, or the proceeds thereof, including the pending lawsuit between Sellers and GeoQuest, a division of Schlumberger N.V.
- (g) Permits. All governmental licenses, permits, variances, consents and approvals specifically associated or used in connection with the ownership of the Assets and operation of the Non-Mexican Business ("Governmental Permits"), including those listed on Exhibit 2.1(g).
- (h) Licenses. All rights under licenses to or from third parties specifically associated or used in connection with the Non-Mexican Business (the "Licenses"), including those listed on Exhibit 2.1(h).

- (i) Assigned Agreements. All rights under the contracts, agreements and commitments related to the Non-Mexican Business (the "Assigned Agreements"), including those listed on Exhibit 2.1(i).
- (j) Subsidiaries. All of the issued and outstanding capital shares of the Mexican Subsidiaries (collectively the "Shares"), which own all assets and employ all employees of the Business located in Mexico.
- (k) Other. All: (i) interests in any partnership or joint venture exclusively associated or used in connection with the Business; (ii) rights associated with any Liability to be assumed by Buyer under this Agreement; (iii) restrictive covenants and other obligations of present employees of the Business and (iv) deposits and prepaid expenses specifically associated with the Business, including those prepaid expenses listed on Exhibit 2.1(k).
- **2.2** Excluded Assets. The Purchased Assets shall not include the following assets of Sellers or their Affiliates (the "Excluded Assets"):
 - (a) Except for cash in the amount of \$350,000 held by the Mexican Subsidiaries on the Closing Date, cash and cash equivalents (other than the Receivables and deposits included as Purchased Assets):
 - (b) rights of any Seller Party under this Agreement;
 - (c) Tax records and returns of any Seller Party or their Affiliates, other than those of the Mexican Subsidiaries;
 - (d) all of the rights, claims and interests of any Seller Party or their Affiliates in the Core Lab Marks;
 - (e) all of the rights, claims or causes of action of any Seller Party or any of their Affiliates (other than the Mexican Subsidiaries), against third parties to the extent they relate to litigation associated with the Coherence Cube Patents or other Excluded Assets pending as of the Closing Date;
 - (f) any rights of any Seller Party or their Affiliates under its insurance policies to the extent that such right relates to an Excluded Asset or an Excluded Liability, <u>provided</u> that insurance policies solely owned by the Mexican Subsidiaries shall continue to be owned by them;
 - (g) personal property items owned by employees of the Business that are listed on Exhibit 2.2;
 - (h) computer systems, software licenses and other Intellectual Property that are listed on Exhibit 2.2;

- (i) any of the assets of any Seller Party or their Affiliates not primarily used in the conduct of the Business, including the Sellers' shared accounting software and hardware and e-mail servers;
- (j) ownership of the Coherence Cube Patents (except as to rights licensed to Buyers under the License Agreement);
- (k) all refunds of Tax due or to become due to the Mexican Subsidiaries or any Seller with regard to the Business that are attributable to Tax periods or portions thereof ending before the Closing Date; and
 - (l) all other assets listed on Exhibit 2.2.
- 2.3 Purchase Price. The purchase price (the "Purchase Price") for the Purchased Assets shall be \$18,433,000, plus the assumption by Buyers of the Assumed Liabilities in accordance with Section 2.5. The Purchase Price shall be subject to adjustment as set forth in Section 2.4 below.
- 2.4 Purchase Price Adjustments. The Purchase Price is subject to adjustments to be determined as follows:
 - (a) For purposes of this Agreement, the term "March 31, 2004 Net Assets Statement" shall mean that statement attached as Exhibit 2.4(a) hereto, which has been compiled from the books of the Sellers and the Mexican Subsidiaries in accordance with their usual accounting methods and practices.
 - (b) Within sixty (60) days after the Closing Date, the Seller Parties and the Buyers shall cause PricewaterhouseCoopers LLP ("PwC") to perform an audit (the "Audit") of the financial statements constructed from the operating financial statements of the Business for the year ended December 31, 2003. Such Audit will be conducted in accordance with generally accepted auditing standards of the United States and be consistent with the audit procedures followed by PwC in their audit of the Seller Parent, and will include the statements of income and cash flows for the year ended December 31, 2003, as well as the Statement of Net Assets as of December 31, 2003. The Seller Parties agree to use their Commercially Reasonable Efforts to assist Buyer in the Audit and to cooperate with the Audit by providing access to records and personnel. The Buyer is responsible for the fees, costs and expenses of PwC and any support required by PwC.
 - (c) Within 15 days after delivery of the Audit, PwC will prepare a Statement of Net Assets as of the Closing Date, taking into account the results of the Audit and using the same standards. PwC shall issue a report thereon (the "Report") which sets forth the net asset adjustments (calculated at 90% of the gross asset adjustments except for the Pemex receivables which shall be calculated at 70%) to those items on the March 31, 2004 Net Assets Statement (and only those items) identified with an asterisk(*) needed to reflect the difference in net book value of those items from the balances on the March 31, 2004 Net Asset Statement to and including the balances of those items as of the Closing Date (the "Closing Date Adjustments"). PwC shall be instructed to permit the Parties and their Representatives, at the earliest practicable date, access to and copies

5

of the work papers and calculations related thereto. The fees, costs and expenses related to PwC in connection with the preparation of the Report shall be shared equally by the Buyers and the Seller Parties.

- (d) Any dispute which may arise between the Buyers and Seller Parties as to the Closing Date Adjustments shall be resolved in the following manner:
 - (i) If either the Buyers or the Seller Parties disputes the calculation of the Closing Date Adjustments or any portion thereof, the disputing Party shall notify PwC and the other Parties within thirty (30) days after receipt of PwC's report, and shall specify therein in reasonable detail the basis and reason for such dispute and the amount which is in dispute; and
 - (ii) PwC shall resolve the matter, and shall be instructed to do so within thirty (30) days of the submission to it of the dispute and, in any case, as soon as practicable after such submission. The fees, costs and expenses related to PwC in connection with a dispute regarding the calculation of the Closing Date Adjustments or any portion thereof pursuant to this Section 2.4(d) shall be shared equally by the Buyers and the Seller Parties. This provision for PwC shall be specifically enforceable by the parties and the decision of PwC in accordance herewith shall be final and binding and there shall be no right of appeal therefrom.
- (e) Within ten (10) days after the later of (i) the delivery of PwC's Report, or (ii) the date of the decision of PwC in connection with any dispute made in accordance with the provisions of Section 2.4(c)(ii) above:
 - (i) If the aggregate net amount of the Closing Date Adjustments is a positive number, the Buyers shall pay to the Seller Parties, by wire transfer of immediately available funds, to such bank as indicated by the Seller Parties, the amount of such Closing Date Adjustments; and
 - (ii) If the aggregate net amount of the Closing Date Adjustments is a negative number, the Seller Parties shall pay to the Buyers, by wire transfer of immediately available funds, to such bank as indicated by the Buyers, the amount of such Closing Date Adjustments.

2.5 Assumed Liabilities; Excluded Liabilities.

(a) On the Closing Date, the Buyers and the Sellers shall execute and deliver an assignment and assumption agreement in the form set forth in Annex 2 (the "Assignment and Assumption Agreement") pursuant to which the Buyers, allocated among them as they may determine, shall assume and agree to pay, perform or otherwise discharge, in accordance with their respective terms and subject to the respective

conditions thereof and subject further to the provisions of <u>Section 2.5(b)</u> (collectively, the "Assumed Liabilities"):

- (i) all of the Liabilities and obligations of any Seller accruing after the close of business on the Closing Date (and not attributable to any violation, breach or failure to perform occurring prior to the Closing) under the Assigned Agreements and under any licenses of Intellectual Property, Governmental Permits and Licenses included in the Purchased Assets (other than any of the foregoing required to be listed in Section 3.15, 3.16 or 3.17 of Sellers' Disclosure Schedule and not so listed, provided that to the extent Buyers receive any material benefit from any of the foregoing not so listed, the Buyers shall assume a proportionate liability therefor up to the amount of the benefits so received);
- (ii) any Liability or obligation of any Seller (including the obligations to make payments) resulting from facts or circumstances arising after the Closing Date (and not attributable to facts or circumstances occurring prior to the Closing) under all licenses, permits, approvals, certificates of occupancy and operating rights included in the Purchased Assets (other than any of the foregoing required to be listed in Section 3.15, 3.16 or 3.17 of Sellers' Disclosure Schedule and not so listed); and
- (iii) any other Liabilities of any Seller set forth on Exhibit 2.5.
- (b) Other than the Assumed Liabilities, no Buyer shall assume or be obligated to pay, perform or otherwise assume or discharge, and the Seller Parties shall indemnify the Buyers and their Representatives and hold them harmless from, any obligations or Liabilities of any Seller Party or their Affiliates (the "Excluded Liabilities"), including any of the following:
 - (i) any Liabilities or obligations in respect of the Excluded Assets;
 - (ii) except as provided in <u>Section 6.5(d)</u>, any Liabilities or obligations of any Seller Party or their Affiliates to indemnify their respective officers, directors, employees, agents or Affiliates, or any individual serving in such a capacity with a third party at the request of any Seller Party or their Affiliates or any individual serving in such a capacity with a third party at the request of any Seller Party or their Affiliates;
 - (iii) subject to <u>Section 2.9</u>, any Taxes imposed on (A) any Seller Party or their Affiliates (other than the Mexican Subsidiaries), whether or not assessed or determined on, before or after the Closing Date, and (B) the Mexican Subsidiaries with respect to periods ending before the Closing Date; and

- (iv) any Liability or obligation of any Seller Party or their Affiliates with respect to employment or consulting agreements (except the Employment Agreements to the extent provided in Section 6.5(a)), Employee Benefit Plans, or amounts owing on or prior to the Closing Date for commissions or compensation, termination, severance or other payments to present or former employees, officers or directors of any Seller Party or their Affiliates.
- 2.6 Allocation of Purchase Price. The Purchase Price shall be allocated among the Purchased Assets in accordance with Section 1060 of the Code and the regulations promulgated thereunder. Such allocation shall be set forth on Exhibit 2.6 hereto. The allocation of Purchase Price may change after the Closing Date. Any change to the Purchase Price allocation following the Closing Date shall be as agreed to in good faith by the Parties. The Buyers agree to use the adjusted allocation of Purchase Price for purposes of Section 1060 of the Code as long as such allocation represents the fair market value of the Purchased Assets.
- 2.7 The Closing. The closing of the transactions contemplated by this Agreement (the "Closing") is taking place concurrently with the execution and delivery of this Agreement and the Transaction Documents, at 9:00 a.m. local time, on April 22, 2004 (the "Closing Date") at such location as the Parties have agreed. For accounting purposes the Closing shall be deemed to have occurred at 11:59 p.m. on the Closing Date.

2.8 Deliveries at the Closing. At the Closing:

- (a) the Sellers are delivering to the Buyer the following:
 - (i) duly executed bills of sale, assignments, and other documents or instruments of conveyance, transfer or assignment (including separate documents or instruments with respect to Purchased Assets located in Canada as may be deemed advisable by the Parties) as are necessary or appropriate to vest or confirm in Buyers all right, title and interest in and to all of the Purchased Assets;
 - (ii) the Assignment and Assumption Agreement, executed by Sellers;
 - (iii) License Agreement in the form of <u>Annex 3</u> hereto (the "License Agreement") with respect to the technology represented by the Coherence Cube Patents (including the name Coherence Cube), executed by Core Lab Global;
 - (iv) Sublease Agreement in the form of Annex 4 hereto (the "Sublease Agreement") with respect to certain space in the Leased Real Property to be subleased by the appropriate Seller to the appropriate Buyer, executed by the appropriate Seller subject to the landlord's consent, which the Parties shall use Commercially Reasonable Efforts to obtain;

HOU:2284955.16

- (v) an instrument reasonably satisfactory to the Buyers' counsel appointing each Buyer as the attorney-in-fact for each Seller Party to institute and prosecute (in that Buyer's own name or in the name of the appropriate Seller Party but for the benefit, and at the expense, of Buyers) any proceedings deemed necessary or appropriate by any Buyer to collect, assert or enforce its right, title and interest in and to the Purchased Assets and the Business:
- (vi) a certificate of the Secretary or Assistant Secretary of each Seller Party, dated as of the Closing Date, certifying as to (A) a copy of corporate resolutions authorizing the execution and delivery of the Transaction Documents to which it is or is to become a party and the consummation of the transactions contemplated by the Transaction Documents, and (B) incumbency of the officers and other personnel of that Seller Party executing this Agreement and any certificate, instrument or document to be delivered by that Seller Party at the Closing;
- (vii) (x) written resignations of all of the incumbent directors, statutory examiners or persons holding similar positions of the Mexican Subsidiaries, which shall include a waiver of all present and future indemnification and claims against the Mexican Subsidiaries, including with respect to any compensation, profit sharing, stock option, stock warrant or other arrangements, if any, (y) election of directors designated by Buyers, and (z) revocation of powers of attorney in favor of employees of Seller Parties and their Affiliates (other than the Mexican Subsidiaries);
- (viii) certificates, if any, representing all of the Shares duly endorsed or accompanied by duly executed assignment documents, or other documents of conveyance or evidence of transfer of the Shares to the appropriate Buyers and/or their designees that the Buyers may reasonably specify;
- (ix) evidence of meetings of shareholders of the Mexican Subsidiaries at which all necessary action has been taken with respect to authorization of the transactions provided for herein;
- (x) consents of such counterparties to those contracts, licenses or leases listed on Exhibit 1(b) as have been obtained on or prior to the Closing Date; and
- (xi) such further certificates, instruments and other documents as reasonably shall be requested by the Buyers' counsel to effect the transaction contemplated by this Agreement.
- (b) the Buyers are delivering to the Sellers the following:

- (i) the Purchase Price, as provided in <u>Section 2.3</u>, for the account of the Sellers, by wire transfer of immediately available funds in the amount of the Purchase Price, to the account or accounts specified in writing by the Sellers to the Buyers on or before the second Business Day prior to the Closing:
- (ii) the amount of \$250,000, in payment of the initial nonrefundable fee under the License Agreement, by wire transfer of immediately available funds to the account specified in writing by or on behalf of Core Lab Global to the Buyers on or before the second Business Day prior to the Closing;
- (iii) the Assignment and Assumption Agreement, and all other appropriate instruments of assumption and other documents or instruments as shall be necessary to cause Buyer to assume any Liabilities, including any Employment Agreement, to be assumed by Buyers hereunder, executed by Buyers;
- (iv) the License Agreement, executed by the appropriate Buyer;
- (v) the Sublease Agreement, executed by the appropriate Buyer subject to the landlord's consent;
- (vi) a certificate of the Secretary or Assistant Secretary of each Buyer, dated as of the Closing Date, certifying as to (A) a copy of corporate resolutions authorizing the execution and delivery of the Transaction Documents to which it is or is to become a party and the consummation of the transactions contemplated by the Transaction Documents, and (B) incumbency of the officers and other personnel of that Buyer executing this Agreement and any certificate, instrument or document to be delivered by that Buyer at the Closing; and
- (vii) such further certificates, instruments, and other documents as reasonably shall be requested by the Seller Parties' counsel to effect the transaction contemplated by this Agreement.
- 2.9 Prorations. At or after the Closing, (a) Taxes that are otherwise calculated on an annual basis such as real and personal property Taxes, water charges, sewer rents and other utility charges and rental deposits in respect of the Purchased Assets and the assets of the Mexican Subsidiaries, shall be prorated on a per diem basis as of the Closing Date and (b) Taxes of the Mexican Subsidiaries based on net or gross income, shall be apportioned as of the Closing Date using a closing-of-the books method, in each case with the applicable Sellers being responsible for such items relative to periods or portions thereof prior to the Closing Date and the applicable Buyers being responsible for such items relative to periods or portions thereof commencing on or subsequent to the Closing Date. To the extent any such prorations are not settled at the Closing, the Party assessed any such amount shall, promptly upon determining the

amount to be prorated, invoice the other Party for the prorated portion thereof to be borne by such other Party, and such other Party shall pay such amount promptly upon receipt thereof.

Nonassignable Items. Notwithstanding anything in this Agreement to the contrary, to the extent that the assignment of all or any portion of any Purchased Asset listed on Exhibit 1(b) shall be prohibited by Law or the terms thereof or require the consent of the other party thereto (if any) or any other third party, (a) this Agreement shall not constitute an agreement to assign any that Purchased Asset (each, a "Nonassignable Item"), and (b) no breach of this Agreement shall have occurred by virtue of that nonassignment. In that event, Sellers after the Closing, will, at the request of the appropriate Buyers but without any obligation to incur any material cost, expense or Liability, (x) use their Commercially Reasonable Efforts to obtain as promptly as practicable the consents required to permit assignment of the Nonassignable Items, and effect such assignment if and when consent is obtained, (y) at the request of Buyer, operate and maintain the Nonassignable Assets as agent for, and as directed by, the appropriate Buyer utilizing the personnel and funds of the appropriate Buyer (provided that such Buyer shall hold the Sellers harmless for any Liabilities incurred in connection with such operation and maintenance), provide the appropriate Buyer with the benefits of, and preserve for the benefit of Buyer the rights of Sellers under, such Nonassignable Items, and (z) hold for the benefit of, and immediately deliver to, the appropriate Buyer any and all revenues received on account of the Nonassignable Assets; provided, however, that nothing in this Section 2.10 shall (i) require any Seller to make any material expenditure or incur any material obligation on its own or on any Buyer's behalf, or (ii) prohibit any Seller from ceasing all operations and business activities of the Business at and following the Closing.

3. REPRESENTATIONS AND WARRANTIES OF SELLER PARTIES.

Each Seller Party represents and warrants, severally, as to such Seller Party, to each Buyer that the statements contained in this <u>Section 3</u>, or in any certificate delivered by any Seller Party, as to such Seller Party are correct and complete as of the date of this Agreement and the Closing Date, except as set forth in the portion of the disclosure schedule of the Sellers delivered to the Buyers on the execution and delivery of this Agreement (the "Sellers' Disclosure Schedule") that refers to the particular provision of this <u>Section 3</u>.

3.1 Organization and Qualification of the Seller Parties. Seller Parent is a public limited liability company validly existing and in good standing under the laws of the Netherlands. Each Seller is a company or limited partnership validly existing and in good standing under the laws of its jurisdiction of organization. Core Lab LP is duly qualified to do business as a foreign limited partnership in Texas.

3.2 Authorization of Transaction.

- (a) Each Seller Party has full corporate or partnership power and authority to execute and deliver, and to perform its obligations under, each Transaction Document to which it is or is to become a party.
- (b) Each Seller Party has duly authorized the execution, delivery and performance of each Transaction Document to which it is or is to become a party.

- (c) Each Transaction Document has been or will be duly executed and delivered by each Seller Party that is or is to become a party to that Transaction Document and constitutes, or on execution and delivery will constitute, the valid and legally binding obligation of that Seller Party, in each case enforceable against that Seller Party in accordance with its respective terms and conditions, subject, however, to the effects of bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting creditor's rights generally and to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- 3.3 Noncontravention. Subject to obtaining or making all of the Sellers' Required Consents, neither the execution, delivery or performance of any Transaction Document by any Seller Party, nor the consummation of the transactions they contemplate, will:
 - (a) Violate any Law, order, judgment or decree applicable to any Seller Party or their Affiliates, as to the Purchased Assets, the Assumed Liabilities or the Mexican Subsidiaries or their assets or Liabilities, except where the violation would not, individually or in the aggregate, have a Material Adverse Effect or a material adverse effect on the ability of the Seller Parties to consummate the transactions contemplated by the Transaction Documents;
 - (b) Violate any provision of the Articles of Association, limited partnership agreement or other corporate governing documents of any Seller Party or of the Mexican Subsidiaries; or
 - (c) Except as to Nonassignable Items, conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, or require any notice under any material agreement, contract, lease, license, instrument or other arrangement to which any Seller Party or their Affiliates (including the Mexican Subsidiaries), is a party or by which it is bound, or to which any of its assets (including the Purchased Assets), the Assumed Liabilities or the assets and Liabilities of the Mexican Subsidiaries is subject, except where the violation, conflict, breach, default, acceleration, termination, modification, cancellation or failure to give notice would not, individually or in the aggregate, have a Material Adverse Effect or a material adverse effect on the ability of the Seller Parties to consummate the transactions contemplated by the Transaction Documents.
- 3.4 Consents. None of the Seller Parties and their Affiliates (including the Mexican Subsidiaries) is required to give any notice to, make any filing with, or obtain any authorization, consent or approval of, any Person (including any Governmental Authority) in order to consummate the transactions contemplated by the Transaction Documents, except for the Sellers' Required Consents, and except for notices, filings, authorizations, consents or approvals that, if not made or obtained, would not have a Material Adverse Effect or a material adverse effect on the ability of the Seller Parties to consummate the transactions contemplated by the Transaction Documents.

3.5 Purchased Assets.

- (a) The Sellers among them have good and valid title to all of the Purchased Assets, free and clear of all Encumbrances (other than Permitted Encumbrances). At Closing, the Sellers will convey to the Buyers good and valid title to all of the Purchased Assets, free and clear of all Encumbrances (other than Permitted Encumbrances and any Encumbrances created by Buyers).
- (b) All Receivables and all accounts receivable of AGI Mexicana represent sales of goods or services actually made in the Ordinary Course of Business and to the Sellers' Knowledge are not disputed by the payor, other than the Specified Receivable. No Receivables or accounts receivable of AGI Mexicana have been sold or factored.
- (c) Exhibit 2.1(b) summarizes all work currently in process included in the Purchased Assets. Other than work in process of AGI Mexicana, all such work is being performing under written contracts and is reasonably expected by Sellers to be fully chargeable in the amounts indicated at the times provided in those written contracts.
- (d) The Purchased Assets include all assets required by Sellers and the Mexican Subsidiaries to provide to third parties the products and services of the Business as currently provided, except to the extent of assets used in providing general corporate services customarily provided to Sellers by their Affiliates, including telephone, accounting, insurance, Tax, and employee benefits processing.
- 3.6 Companies Involved with the Business. The Sellers and the Mexican Subsidiaries are the only Persons that conduct any material part of the Business or own any of the Purchased Assets.

3.7 Tax Matters.

- (a) The Seller Parties and their Affiliates (including the Mexican Subsidiaries) have filed all required Tax Returns with respect to the Business, the Purchased Assets, the Assumed Liabilities, the Mexican Subsidiaries and its assets and Liabilities and have paid all Taxes shown on such Tax Returns as owing, except where the failure to file Tax Returns or to pay Taxes would not, individually or in the aggregate, have a Material Adverse Effect. All such Tax Returns were correct and complete in all material respects at the time of filing.
- (b) All Taxes owed by, and all Tax deficiencies asserted or assessed against, any Seller Party or any of their Affiliates with respect to the Business, the Purchased Assets and the Assumed Liabilities, or owed by or asserted or assessed against, the Mexican Subsidiaries, have been resolved and paid in full.
- (c) None of the Seller Parties and their Affiliates (including the Mexican Subsidiaries) has taken any action to waive any statute of limitations in respect of Taxes applicable to the Business, any Purchased Asset or any Assumed Liability or the Mexican Subsidiaries or their assets or Liabilities or agreed to any extension of time with respect to a Tax assessment or deficiency applicable to the foregoing.

3.8 Leased Real Property. Exhibit 2.1(d) lists and describes all Leased Real Property and the applicable leases and subleases. None of the Seller Parties, their Affiliates and, to the Sellers' Knowledge, any other Person is in breach of or default under, and no event has occurred that, with notice or lapse of time, would constitute a breach or default or permit termination, modification or acceleration under, any leases or subleases for the Leased Real Property, except for such breaches, defaults or events as to which requisite waivers have been obtained or that would not, individually or in the aggregate, have a Material Adverse Effect. Seller has made available to the Buyer a correct and complete copy of each lease or sublease listed on Exhibit 2.1(d). Except for the Leased Real Property, no real property is used in the operation of the Business, other than incidental use of other real property owned or leased by the Seller Parties and their Affiliates.

3.9 Legal Compliance.

- (a) There are and since April 15, 1999 have been no violations of any Law (other than any Tax Laws or Environmental Laws, which are addressed in <u>Sections 3.7</u> and <u>3.10</u>, respectively) applicable to the Business, the Purchased Assets, the Assumed Liabilities or the Mexican Subsidiaries or their assets or Liabilities, except for violations that would not, individually or in the aggregate, have a Material Adverse Effect or have been cured in the manner provided by applicable Mexican Law.
- (b) Without limiting the generality of the foregoing, the Seller Parties and their Affiliates (including the Mexican Subsidiaries) and their Representatives, in connection with the Business, the Purchased Assets, the Assumed Liabilities and the Mexican Subsidiaries and their assets and Liabilities, have (i) conducted their activities in full compliance with the provisions of the U.S. Foreign Corrupt Practices Act and all other applicable Laws regarding corruption and commercial bribery, (ii) not had any material unrecorded fund or asset for any purposes, (iii) not accumulated any material amount of funds or cash without properly recording them in the appropriate Person's books and records, (iv) not made any material payment other than for the purposes recorded in their books and records, and (v) to Sellers' Knowledge, not made, directly or indirectly, any other illegal contribution, gift, bribe, rebate, payoff, influence payment, kickback, or other payment to any Person, public or private, regardless of form (including any non-cash consideration), to obtain favorable treatment or personal or commercial advantage, or accepted any such payment.

3.10 Environmental Matters.

(a) Each of Seller Parties and their Affiliates (including the Mexican Subsidiaries), the Purchased Assets, the Assumed Liabilities and the Mexican Subsidiaries and their assets and Liabilities is in compliance with applicable Environmental Laws with respect to the Non-Mexican Business, the Purchased Assets, the Assumed Liabilities and the Mexican Subsidiaries and their assets and Liabilities, except where the non-compliance would not, individually or in the aggregate, have a Material Adverse Effect.

14

- (b) There are no pending, or to the Sellers' Knowledge, threatened claims, demands, actions, notices of noncompliance, administrative proceedings, lawsuits or investigations, and there are no outstanding orders, consent decrees or judgments with respect to the Non-Mexican Business, the Purchased Assets and the Assumed Liabilities or against the Mexican Subsidiaries under any Environmental Laws, claiming any personal injury, property damage or other harm allegedly resulting from the Release of or exposure to Hazardous Substances, or arising from any activities not in compliance with any Environmental Laws, except for such matters that would not, individually or in the aggregate, have a Material Adverse Effect.
- (c) None of the Seller Parties and their Affiliates (including the Mexican Subsidiaries) has received any written notice or request for information that, with respect to the Non-Mexican Business, the Purchased Assets, the Assumed Liabilities or the Mexican Subsidiaries or their assets or Liabilities, it is a potentially responsible party under CERCLA or any similar state or non-U.S. Law, except for requests or notices with respect to such matters that would not, individually or in the aggregate, have a Material Adverse Effect, and has not received any written notice of any Governmental Authority's allegation or investigation of any criminal violations of any Environmental Laws with respect to the Non-Mexican Business, the Purchased Assets, the Assumed Liabilities or the Mexican Subsidiaries or their assets or Liabilities.
- 3.11 Powers of Attorney. There are no outstanding powers of attorney executed by or on behalf of (a) the Mexican Subsidiaries or (b) any Seller Party or any of their Affiliates with respect to any of the Non-Mexican Business, the Purchased Assets and the Assumed Liabilities.
- 3.12 Litigation. There is no claim, action, proceeding or investigation pending or, to the Sellers' Knowledge, threatened against the Mexican Subsidiaries or relating to the Non-Mexican Business, the Purchased Assets or the Assumed Liabilities before any Governmental Authority, or any judgment, decree or order of any Governmental Authority, that would, individually or in the aggregate, have a Material Adverse Effect or that would delay or prevent the consummation of the transactions contemplated by the Transaction Documents.

3.13 The Mexican Subsidiaries.

- (a) Each of the Mexican Subsidiaries is a Mexican sociedad anonima de capital variable validly existing and in good standing under the laws of Mexico.
- (b) Except as set forth on <u>Section 3.13</u> of Sellers' Disclosure Schedule, the only activities conducted, assets owned or leased or Liabilities incurred or outstanding by or involving AGI Mexicana relate to the Business, and the only activities conducted by, assets owned or leased or Liabilities incurred by ST relate to providing employee services to AGI Mexicana. Neither of the Mexican Subsidiaries owns any equity or other securities or options, warrants, convertible securities or other rights of any kind to acquire any equity interest, of any other Person, regardless of whether constituting an Affiliate of the Mexican Subsidiaries, and is not a partner, venturer or participant in any partnership, joint venture or other Person.

15

- (c) Section 3.13 of the Sellers' Disclosure Schedule lists (i) all of the assets having a net book value of more than \$10,000, (ii) each Liability (other than obligations of performance under contracts) exceeding \$10,000, (iii) all written contracts, agreements, leases, subleases and other legally binding commitments that individually provide for or involve payment by or to either of the Mexican Subsidiaries, including all performance bonds (the "Mexican Agreements"), (iv) all material Governmental Permits, (v) all material Licenses, and (vi) all licenses for the use of Intellectual Property, of each of the Mexican Subsidiaries. The Liabilities of the Mexican Subsidiaries not listed in Section 3.13 of the Seller's Disclosure Schedule in the aggregate do not exceed \$50,000. The Mexican Subsidiaries will have no obligation under performance bonds attributable to periods before the Closing Date for actions taken prior to the Closing Date. The provisions of the foregoing sentence are not subject to the limitations in Section 8.5. Each of the Mexican Subsidiaries has good and valid title to, or where specified a valid leasehold interest in, all such assets, free and clear of all Encumbrances (other than Permitted Encumbrances). The representations and warranties in Section 3.8 are true with respect to all premises occupied by the Mexican Subsidiaries as if those premises were Leased Real Property, and the representations and warranties in Section 3.16 are true with respect to the Mexican Agreements mutatis mutandis.
- (d) AGI Mexicana's authorized share capital consists of (i) 50,000 shares of fixed capital with a nominal value of 1 Mexican peso each, of which 49,999 shares are issued and outstanding and registered in the name of Core Lab de Mexico, and 1 share is issued and outstanding and registered in the name of Core Laboratories International B.V., and (ii) 96,869,242 shares of variable capital, with a nominal value of 1 Mexican peso each, all of which shares are issued and outstanding and registered in the name of Core Lab de Mexico, subject in each case to no Encumbrances. There are outstanding no options, warrants, convertible securities or other rights of any kind to acquire any equity interest in AGI Mexicana.
- (e) ST's authorized share capital consists of 50,000 shares of fixed capital with a nominal value of 1 Mexican peso each, of which 49,999 shares are issued and outstanding and registered in the name of Core Lab de Mexico and 1 share is issued and outstanding and registered in the name of Core Laboratories International B.V., subject in each case to no Encumbrances. There are outstanding no options, warrants, convertible securities or other rights of any kind to acquire any equity interest in ST.
- (f) Since December 11, 1998, the minute books and corporate records of each of the Mexican Subsidiaries have been maintained in the Ordinary Course of Business and in accordance with applicable Law, are complete and correct in all material respects, and contain accurate and complete records of all actions taken by their respective shareholders, boards of directors and directors' committees.

3.14 Employee Matters.

(a) Section 3.14 of the Sellers' Disclosure Schedule sets forth a list of all current Employees and the Persons employing them, with their base annual salary and latest year's bonus, accrued vacation, and all currently effective written agreements of

employment between any Seller Party or its Affiliates (including the Mexican Subsidiaries) and any Employee (the "Employment Agreements") and all bonus and commission plans applicable to any Employee, which together with the Employee Benefit Plans include all obligations to Employees (including any collective bargaining, bonus, profit sharing, thrift, compensation, stock option, restricted stock, pension, retirement, deferred compensation, employment, termination, severance or other plan, agreement, trust, fund, policy or arrangement) other than those arising solely as a matter of Law. Copies of all employment agreements, Employee Benefit Plans and written personnel policies and procedures regarding any Employee have been made available to the Buyers.

- (b) No Employee is subject to a collective bargaining or similar agreement or represented by a labor union or organization, and to the Sellers' Knowledge no efforts currently are under way to subject any Employees to any such agreement or to have any union certified with respect to any of the Employees or the Business. There have not been any organized work stoppages or strikes in the Business and, to the Sellers' Knowledge, no such work stoppage or strike is threatened or imminent.
- (c) Each of the Seller Parties and their Affiliates (including the Mexican Subsidiaries) has complied with all provisions of applicable Law, including ERISA and the Code, pertaining to the Employees, including such Laws relating to labor relations, profit-sharing, equal employment and fair employment practices, working hours and conditions, occupational safety and health, except for any noncompliance that would not, individually or in the aggregate, have a Material Adverse Effect. There are no actions, proceedings or charges pending or, to the Sellers' Knowledge, threatened, involving any Employee, Employee Benefit Plans, the Business, or the Mexican Subsidiaries, before any Governmental Authority having jurisdiction over any such matters.
- (d) Section 3.14(d) of the Sellers' Disclosure Schedule lists each Employee Benefit Plan that any Seller Party or any of their Affiliates (including the Mexican Subsidiaries) sponsors, maintains, or contributes to for the benefit of any Employee (collectively, the "Employee Plans"). Each of the Employee Plans that is intended to meet the requirements of Section 401(a) of the Code and that provides for eligible rollover contributions described in Section 402(f)(2)(A) of the Code complies with the applicable requirements of ERISA, the Code and other applicable Laws, except where the non-compliance would not, individually or in the aggregate, have a Material Adverse Effect. All contributions (including all employer contributions and employee salary reduction contributions) that are due have been paid to each Employee Plan that is any Employee Pension Plan that is subject to Section 302 of Title I of ERISA and Title IV of ERISA as well as Section 412 of the Code, and all contributions for any period ending on or before the Closing Date that are not yet due have been paid to any Employee Pension Benefit Plan which is subject to Section 302 of Title I of ERISA and Title IV of ERISA as well as Section 412 of the Code, or accrued in accordance with the past custom and practice of the Seller Parties, and no liability under Title IV of ERISA or Section 302 of ERISA has been incurred by any Seller Party that would become a Liability for any Buyer.

(e) No Employee Benefit Plan or Employee Agreement provides medical, surgical, hospitalization, death or similar benefits (whether or not insured) for employees or former employees of any of the Seller Party or any subsidiary for periods extending beyond their retirement or other termination of service.

3.15 Intellectual Property.

- (a) Exhibit 2.1(f) sets forth a true and complete list of all Intellectual Property, which includes all patents and patent applications, trademark registrations and applications, service mark registrations and applications, seismic processing software, copyright registrations and applications, used or held for use in connection with the Business or required for the conduct of the business of the Business in the Ordinary Course of Business, showing whether such are owned or licensed.
- (b) The Sellers or the Mexican Subsidiaries are the sole and exclusive owners of all Intellectual Property listed as owned on Exhibit 2.1(f), free and clear of all Encumbrances, other than Permitted Encumbrances.
- (c) To the Sellers' Knowledge, there are no conflicts with or infringements of any Intellectual Property by any third party. To the Sellers' Knowledge, the conduct of the Businesses does not conflict with or infringe on or misappropriate in any way any proprietary right of any third party. There is no claim, suit, action or proceeding pending or, to the Sellers' Knowledge, threatened against the Sellers or the Mexican Subsidiaries (i) alleging any such conflict with or infringement misappropriation of any third party's proprietary rights or (ii) challenging the ownership, use, validity or enforceability of the Intellectual Property.
- (d) The computer software necessary for the conduct of the Business as conducted on the Closing Date was either (i) developed by employees of the Seller Parties and their Affiliates (including the Mexican Subsidiaries) within the scope of their employment, (ii) developed on behalf of a Seller Party or one or more of their Affiliates (including the Mexican Subsidiaries) by a third party, and all ownership rights therein have been assigned or otherwise transferred to or vested in the Seller Party or one or more of their Affiliates pursuant to written agreements with such employees or third parties, or (iii) licensed or acquired from a third party pursuant to a written license, assignment, or other contract that is included in the Purchased Assets and is in full force and effect and of which none of the Seller Parties and their Affiliates (including the Mexican Subsidiaries) is in material breach.
- (e) None of the Seller Parties and their Affiliates (including the Mexican Subsidiaries) has entered into any consent, indemnification, forbearance to sue, settlement agreement, exclusive license, or cross-licensing arrangement with any Person related to the Intellectual Property or the intellectual property of any third party, except as contained in any license agreements listed on Exhibit 2.1(f).

- (f) Sellers have taken reasonable steps, consistent with customary industry practice, to protect and preserve its trade secrets. Sellers have made available to Buyers copies of all nondisclosure agreements for Employees.
- (g) None of the Seller Parties and their Affiliates (including the Mexican Subsidiaries) is, or assuming the receipt of Sellers' Required Consents, will be as a result of the execution and delivery of the Transaction Documents or the consummation of the transactions contemplated by them, in breach of any license, sublicense or other agreement relating to the Intellectual Property.
- 3.16 Material Agreements. Section 3.16 of Sellers' Disclosure Schedule lists all written contracts, agreements, leases, subleases and other legally binding commitments related primarily to the Non-Mexican Business that individually provide for or involve payments by or to any Seller Party or their Affiliates of more than \$100,000, or the consequences of a default or termination of which could have a Material Adverse Effect (the "Material Agreements"). None of the Seller Parties and their Affiliates (including the Mexican Subsidiaries) or, to Sellers' Knowledge, any other party to any Material Agreement, is in material breach or default thereunder, and no event has occurred that, with notice or lapse of time, would constitute a material breach or default or permit termination, modification or acceleration, of any Material Agreement, except for such breaches, defaults or events as to which requisite waivers have been, or prior to the Closing will have been, obtained or that would not, individually or in the aggregate, have a Material Adverse Effect. The Seller Parties have not received any written notice of the repudiation, or challenge to the validity, of any Material Agreement by the counterparty thereto. The Seller Parties have made available to the Buyers a correct and complete copy of each Material Agreement required to be listed on Section 3.16.
- 3.17 Governmental Permits. One of the Sellers or the Mexican Subsidiaries holds, and the Sellers and their Affiliates (including the Mexican Subsidiaries) are in compliance with, all Governmental Permits that are required to conduct the business of the Business as currently conducted, except for such failures to hold or comply that would not, individually or in the aggregate, have a Material Adverse Effect. All such Governmental Permits are in full force and effect, and to the Sellers' Knowledge there is no basis for revoking or adversely modifying any such Governmental Permit.
- 3.18 Material Adverse Change. No Material Adverse Change has occurred since December 31, 2003.
- 3.19 Brokers' Fees. None of the Seller Parties and their Affiliates (including the Mexican Subsidiaries) have entered into any arrangement to pay any fees or commissions to any broker, finder or agent with respect to the transactions contemplated by the Transaction Documents for which Buyers and their Affiliates (including the Mexican Subsidiaries after the Closing) could become liable or obligated.

3.20 Accuracy of Information.

- (a) All copies of documents, agreements, and reports in the Due Diligence Materials are accurate and complete copies of the original documents, agreements, and reports.
- (b) The items listed in <u>Section 3.20</u> of the Sellers' Disclosure Schedule, which were part of the Due Diligence Materials, were prepared by the Sellers or their Affiliates in the Ordinary Course of Business for their internal use at the time (or for Buyers' use in connection with the transaction described in this Agreement) and, to the Sellers' Knowledge, are materially accurate taken as a whole, in each case taking into account the time at which and the circumstances in which those items were prepared and subject to the other Due Diligence Materials, including updates of those items.

4. REPRESENTATIONS AND WARRANTIES CONCERNING THE BUYER.

Each Buyer represents and warrants to each Seller Party severally as to such Buyer that the statements contained in this <u>Section 4</u>, and in any certificate delivered by such Buyer at the Closing, as to such Buyer are correct and complete as of the date of this Agreement and the Closing Date, except as set forth in the portion of the disclosure schedule of the Buyers delivered to the Seller Parties on the execution and delivery of this Agreement (the "Buyers' Disclosure Schedule") that refers to the particular provision of this <u>Section 4</u>.

4.1 Organization of the Buyer. Paradigm Geotechnology is a limited liability company validly existing and in good standing under the laws of the Netherlands. Each other Buyer is a company or limited partnership validly existing and in good standing under the laws of its jurisdiction of organization. Each Buyer is duly qualified to do business as a foreign entity in each jurisdiction necessary to perform its obligations under the Transaction Documents.

4.2 Authorization of Transaction.

- (a) Each Buyer has full corporate or partnership power and authority to execute and deliver, and to perform its obligations under, each Transaction Document to which it is or is to become a party.
- (b) Each Buyer has duly authorized the execution, delivery and performance of each Transaction Document to which it is or is to become a party.
- (c) Each Transaction Document has been or will be duly executed and delivered by each Buyer that is or is to become a party to that Transaction Document and constitutes, or on execution and delivery will constitute, the valid and legally binding obligation of that Buyer, in each case enforceable against that Buyer in accordance with its respective terms and conditions, subject, however, to the effects of bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting creditor's rights generally and to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

- **4.3 Noncontravention**. Subject to obtaining or making all of the Buyers' Required Consents, neither the execution, delivery or performance of any Transaction Document by any Buyer, nor the consummation of the transactions they contemplate, will:
 - (a) Violate any Law, order, judgment or decree applicable to any Buyer, except where the violation would not, individually or in the aggregate, have a material adverse effect on the ability of the Buyers to consummate the transactions contemplated by the Transaction Documents;
 - (b) Violate any provision of the Articles of Association, limited partnership agreement or other corporate governing documents of any Buyer; or
 - (c) Conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, or require any notice under any material agreement, contract, lease, license, instrument or other arrangement to which any Buyer is a party or by which it is bound, except where the violation, conflict, breach, default, acceleration, termination, modification, cancellation or failure to give notice would not, individually or in the aggregate, have a material adverse effect on the ability of the Buyers to consummate the transactions contemplated by the Transaction Documents.
- 4.4 Consents. None of the Buyers is required to give any notice to, make any filing with, or obtain any authorization, consent or approval of, any Person (including any Governmental Authority) in order to consummate the transactions contemplated by the Transaction Documents, except for the Buyers' Required Consents, and except for notices, filings, authorizations, consents or approvals that, if not made or obtained, would not have a material adverse effect on the ability of the Buyers to consummate the transactions contemplated by the Transaction Documents.
- **4.5 Brokers' Fees**. None of the Buyers has entered into any arrangement to pay any fees or commissions to any broker, finder or agent with respect to the transactions contemplated by the Transaction Documents for which any Seller Party could become liable or obligated.
- 4.6 Litigation. There is no claim, action, proceeding or investigation pending or, to the Buyers' Knowledge, threatened against any Buyer before any Governmental Authority, or any judgment, decree or order of any Governmental Authority, that would delay or prevent the consummation of the transactions contemplated by the Transaction Documents.

4.7 [Intentionally deleted.]

4.8 Due Diligence Investigation and Other Acknowledgments.

21

(a) The Buyers acknowledge and agree that they have fully conducted and are relying exclusively upon the representations and warranties in this Agreement and their own inspections and investigation in order to satisfy themselves as to the condition and suitability of the Business, assets, real and personal properties, Liabilities, results of operations, condition (financial or otherwise) and prospects of the Business. In addition,

the Buyers acknowledge and agree that they have reviewed all of the Due Diligence Materials to their full and complete satisfaction.

- (b) Except for the representations and warranties set forth in Section 3, the Buyers acknowledge and agree that the Seller Parties and their Affiliates make no representations or warranties (express, implied, at common law, statutory or otherwise) with respect to the accuracy or completeness of the Due Diligence Materials or the Information Memorandum now, previously or hereafter made available to the Buyers in connection with this Agreement (including any description of the Business or the Purchased Assets, market, revenue, price and expense assumptions or environmental information), or of any other information furnished to the Buyers by the Seller Parties or their Representatives.
- (c) The Buyers acknowledge and agree that they have reviewed and accepted the disclaimers set forth in <u>Section 8.8</u> of this Agreement.

5. [INTENTIONALLY DELETED.]

6. POST-CLOSING COVENANTS.

The Parties agree as follows with respect to the period following the Closing.

6.1 General; Certain Assets.

- (a) If after the Closing any further action is necessary to carry out the purposes of the Transaction Documents and the transactions they contemplate, each of the Parties will take such further action (including the execution and delivery of such further instruments and documents) as any other Party not its Affiliate reasonably may request, and the requesting Party will pay all reasonable out-of-pocket costs and expenses of the requested Party (unless the requesting Party is entitled to indemnification for the requested action under <u>Section 8</u>). No Party, however, will be required to take any action that, in the opinion of its counsel, would or could constitute a violation of any Law.
- (b) Without limiting the generality of the foregoing, if requested by Sellers, the Buyers agree to cause the Mexican Subsidiaries to use their Commercially Reasonable Efforts to assist the Sellers and their Affiliates in bifurcating those contracts (the "Consortium Contracts") to which both AGI Mexicana and Core Lab Operations S.A. de C.V. ("Core Lab Operations") are parties and which are listed as Consortium Contracts on Exhibit 6.1 hereto, and otherwise to take reasonable steps in order to continue to provide Core Lab Operations with the benefits of such Consortium Contracts. If any such Consortium Contract shall not be bifurcated as described above, then each Party shall (and shall cause its appropriate Subsidiaries to) act, to the maximum extent permitted by Mexican Law, as the agent or representative of the other Party (or its appropriate Subsidiary) in order to obtain for the other Party the benefits intended to be conferred upon such Party thereunder, and the Parties shall (and shall cause their appropriate Subsidiaries to) cooperate, to the maximum extent permitted by Law and the Consortium Contract with the other in any other reasonable arrangement designed to

provide such benefits to the Party (or Subsidiary) upon whom such benefits are intended to be conferred.

- (c) Sellers and Buyers will use Commercially Reasonable Efforts to cause the Arenque Contract and the contracts giving rise to the accounts receivable of AGI Mexicana listed on Item 5 of Exhibit 6.1 (the "Other Receivables") to be transferred to one or more Persons designated by the Sellers. Pending such transfers, Sellers shall cause their applicable Affiliate to perform all obligations under and pursuant to (i) the Consortium Contracts, (ii) the contract identified in Item 2 of Exhibit 6.1 (the "Arenque Contract"), and (iii) the contracts giving rise to the Other Receivables in accordance with the terms thereof, and shall (and shall cause their applicable Affiliate to) indemnify and hold harmless AGI Mexicana and its Affiliates from and against any Damages arising out of the failure of Sellers or their applicable Affiliate to perform such obligations. If any Party or its Affiliates shall pay or provide any monies on behalf of the other Party or its Affiliates in connection with any Consortium Contract, the Arenque Contract, and the contracts giving rise to the Other Receivables, the Party receiving (or whose Affiliate received) such monies or the benefit thereof shall promptly reimburse the other Party or its Affiliate that paid or provided such monies.
- (d) Prior to any transfer of the Arenque Contract as provided for in Section 6.1(c), Buyers shall cause AGI Mexicana to perform the invoicing and similar ministerial obligations under the Arenque Contract for the benefit of Sellers or Sellers' designee. Sellers shall be responsible for provision of personnel and equipment necessary to perform the services to be rendered to the counterparty under the Arenque Contract and shall reimburse AGI Mexicana for the reasonable out of pocket costs incurred by AGI Mexicana in performing its obligations under the Arenque Contract.
- (e) Sellers and Buyers will use Commercially Reasonable Efforts to cause the leases and domain name listed in items 3 and 4 on Exhibit 6.1 to be transferred to one or more Persons designated by Sellers. Pending such transfer, Sellers shall pay directly, or if not practical, reimburse AGI Mexicana for, any amounts due to the lessors pursuant to the terms of such leases, and AGI Mexicana shall afford to Sellers' designees the use of the facilities covered by such leases.
- (f) Buyers agree that the Other Receivables are for the benefit of Sellers, and unless such Other Receivables shall have been transferred as provided in Section 6.1(c), Buyers shall cause AGI Mexicana promptly upon receipt of any payment on account of such receivables to remit such funds to the account designated by Sellers therefor. AGI Mexicana shall not be obligated to institute litigation or arbitration proceedings or otherwise incur any material out of pocket expense in collection of such receivables unless Sellers shall have agreed to reimburse AGI Mexicana for all such expenses.

6.2 Transition.

(a) Subject to <u>Section 6.1</u>, the Seller Parties will not, and will not permit any their Representatives to, take any action that is designed or intended to have the effect of discouraging any lessor, licensor, customer, supplier or other business associate of Sellers

or their Affiliates (including the Mexican Subsidiaries) with respect to the Business from maintaining the same business relationships with the Buyers (and the Mexican Subsidiaries) with respect to the Business after the Closing as it maintained with respect to the Business prior to the Closing.

- (b) For a period of up to 60 days following the Closing Date, Core Lab LP will provide office space currently occupied by the Employees, heating, ventilating and air conditioning, and utilities for employees of the Business located at Core Lab LP's Houston, Texas facility on a rent-free basis, provided, that long distance telephone service shall be directly billed to the Buyers, or, if provided by Core Lab LP, Buyer will reimburse Core Lab LP for the actual charges occasioned by Buyers. If any employees of the Business continue to occupy any space leased by Core Lab LP after 60 days following the Closing, the Buyers will use their best efforts to relocate such employees promptly and shall pay to Core Lab LP \$36,700 per month thereafter for the space occupied by such employees, provided, that no such holdover shall exceed two months without the written consent of Core Lab LP.
- (c) With respect to the services and assistance provided by any Seller Party under this Section 6.2, (i) the relationship of the Parties shall be solely that of independent contractors, and (ii) Buyers shall indemnify and hold harmless the Seller Parties and their Representatives from and against all Damages for death or injury to persons or property arising out of or in connection with the presence of any Buyer or its Representatives upon the premises of any Seller Party, and whether arising in tort, contract, strict liability, or at law or equity, except through the gross negligence or willful misconduct of any Seller Party or their Affiliates.
- (d) The Buyers shall cause the Mexican Subsidiaries for 60 days following the Closing Date to allow the Sellers' Affiliates use of their T-1 communications line and servers, free of charge.
- After the Closing Date, any asset or any Liability, all other remittances and all mail and other communications that are or otherwise relate to a Purchased Asset or an Assumed Liability and that is or comes into the possession, custody or control of the Sellers (or their Affiliates) shall forthwith be transferred, assigned or conveyed by the appropriate Seller (or its Affiliate) to the appropriate Buyer. Until such transfer, assignment and conveyance, the Sellers (and their respective Affiliates) shall not have any right, title or interest in or obligation or responsibility with respect to such asset or Liability except that the Sellers (and their Affiliates) shall hold any such Purchased Asset in trust for the benefit of Buyers pending such transfer. After the Closing Date, any asset or any Liability, all other remittances and all mail and other communications that are or otherwise relate to an Excluded Asset or an Excluded Liability and that is or comes into the possession, custody or control of the Buyers (or their Affiliates) shall forthwith be transferred, assigned or conveyed by the appropriate Buyer (or its Affiliate) to the appropriate Seller. Until such transfer, assignment and conveyance, the Buyers (and their Affiliates) shall not have any right, title or interest in or obligation or responsibility with respect to such Excluded Asset or Excluded Liability except that the Buyers (and their

Affiliates) shall hold any such Excluded Asset in trust for the benefit of Sellers pending such transfer. The terms of this <u>Section 6.2(e)</u> shall survive the Closing indefinitely.

(f) For a period of up to three (3) months, Sellers will at mutually agreeable good faith prices, provide Buyers such services as Buyers request such as payroll, invoicing, accounts database management and preparation of financial data and such other services as the Parties may reasonably agree.

6.3 Information.

- (a) The Buyers will afford to the Seller Parties and their Representatives access during normal business hours with reasonable notice to the books, records and personnel, and to such other information, and will furnish such cooperation, in each case relating to Business, the Purchased Assets, the Assumed Liabilities and the Mexican Subsidiaries and their assets and Liabilities, as any Seller Party reasonably requests for financial reporting and accounting matters, the preparation and filing of any Tax Returns, the defense of Tax and indemnity claims, and any other purposes related to the Transaction Documents and the transactions they contemplate, except to the extent sharing the information may waive attorney-client or other applicable privileges. In addition, the Sellers will, and the Seller Parent will cause each Seller and its Representatives to, afford the Buyers and their Representatives similar access to any books, records or files retained by the Seller relating to Business, the Purchased Assets, the Assumed Liabilities and the Mexican Subsidiaries and their assets and Liabilities.
- (b) The Seller Parties shall, and shall cause their Representatives to, keep confidential and not use for any purpose (except those described in Section 6.3(a)), any and all information in their possession regarding the Business, the Purchased Assets, the Assumed Liabilities and the Mexican Subsidiaries and their assets and Liabilities, except for disclosures (i) required by Law or any listing or trading agreement concerning its publicly-traded securities (in which case the Seller Parties will use Commercially Reasonable efforts to advise the Buyers prior to the disclosure being made) and (ii) of information already in the public domain through no act or omission of any Seller Party or any of their Representatives.
- 6.4 Transfer Taxes. The Buyers will pay all sales, use, value added, transfer, real property transfer, recording, stock transfer and other similar Taxes and fees, if any, arising out of or in connection with the sale of the Purchased Assets and the consummation of the transactions contemplated by the Transaction Documents, and will indemnify, defend and hold harmless the Sellers and their Representatives with respect to such Taxes. The Buyers will file all necessary documentation and Tax Returns with respect to such Taxes. If required by applicable Law, the applicable Buyer shall pay over to the applicable Seller the amount of any sales, use or value added Tax which may be required by Law to be collected by such Seller, and such Seller shall timely remit such Tax to the appropriate Governmental Authority in compliance with applicable Law.

6.5 Employment Obligations.

- (a) As of the day after the Closing Date, all U.S. Employees and Canadian Employees will become employees of one or more of Buyers and their Affiliates at the same salaries as described in Section 3.14 of the Sellers' Disclosure Schedule. In accordance with the Assignment and Assumption Agreement, Buyers will assume or will cause one of their Affiliates to assume the Seller's obligations under each of the Employment Agreements listed in Section 3.14 of the Sellers' Disclosure Schedule. Except as provided in this Section 6.5 or in Section 6.6, all Liabilities (other than Liabilities attributable to any violation, breach or failure to perform occurring prior to the Closing) in respect of the U.S. Employees and the Canadian Employees arising after the Closing Date will be the responsibility of Buyers or their applicable Affiliates.
- (b) After the Closing, Core Lab LP shall provide continuation coverage under COBRA and any state continuation coverage requirements to all U.S. Employees and their qualified beneficiaries, other than those who elect not to have this coverage. With respect to the period commencing on the Closing Date and ending on June 22, 2004, for each U.S. Employee that elects in writing to accept continuation coverage under COBRA (or if earlier, the date a particular U.S. Employee's employment with Paradigm U.S. terminates), Paradigm US will pay on behalf of such U.S. Employee the cost of this coverage, at the rate in effect for this coverage on the Closing Date; provided, however, that Buyers may charge each U.S. Employee so covered an amount for this period at a rate equal to the rate that U.S. Employee had been charged for health insurance coverage as an employee of Core Lab LP immediately prior to the Closing Date.
- (c) Sellers represent that they and their Affiliates have not discouraged, and agree that they will not (and will not permit their Affiliates to) discourage, any Employee from becoming or remaining an employee of Buyers and their Affiliates (including the Mexican Subsidiaries) following the Closing. After the Closing, Buyers and their Affiliates (including the Mexican Subsidiaries) may determine in their sole discretion which Employees should continue as employees of Buyers and their Affiliates (including the Mexican Subsidiaries). Buyers and their Affiliates may require as a condition of continued employment that the Employee enter into an employment agreement or noncompetition agreement with the applicable Buyer or Affiliate. Any U.S. Employee or Canadian Employee whose employment with any of the Buyers and their Affiliates continues after June 22, 2004 (each a "Continuing Employee") shall be given credit for service with Sellers (and any other entity to the extent credit has heretofore been granted by Sellers) with respect to eligibility for participation and vesting, as applicable, in any welfare benefits plans and other employee benefits plans of the Buyers (including vacation and sick leave plans) where the period or periods of service is or are relevant to the same extent as such service would be credited had it been performed for Buyers; provided, however, that in no event shall the Continuing Employees be entitled to any credit to the extent that it would result in duplication of benefits with respect to the same period of service. In addition, to the extent such actions can be taken without material cost, Buyers shall take appropriate action to cause their welfare benefit plans to waive pre-existing conditions exclusions for Continuing Employees and their dependents and credit amounts paid by Continuing Employees toward deductibles and annual maximums

in the current year of affected welfare benefit plans of Sellers as if such amounts had been paid under Buyers' welfare benefit plans in their years which include the Closing Date. The parties intend that none of the Buyers or their Affiliates shall be responsible for any severance costs owed to any U.S. Employee or Canadian Employee not offered employment with a Buyer or an Affiliate of the Buyers, or who does not accept such an offer, should any of the Seller Parties or their Affiliates terminate that U.S. Employee or Canadian Employee's employment, in which case such severance costs shall remain the obligation of the Seller Parties and their Affiliates to the extent applicable.

- employment with any of the Buyers and their Affiliates (including the Mexican Subsidiaries) terminates for any reason on or prior to June 22, 2004 (each a "Non-Continuing Employee") and the date of that termination. With respect to each Non-Continuing Employee, (i) Sellers shall pay, or shall cause one of their Affiliates to pay, each Non-Continuing Employee the payments that Non-Continuing Employee would have received if instead that Non-Continuing Employee had been terminated as an employee of one of Sellers and their Affiliates as of the Closing Date under the relevant Seller's applicable severance plans and policies in effect immediately prior to the Closing Date, a list of which amounts Sellers have provided to Buyers, and (ii) Buyers shall reimburse Sellers for any portion of the amounts so paid attributable to accrued vacation time to the extent reflected on Section 3.14 of the Sellers' Disclosure Schedule.
- (e) Until the third anniversary of the Closing Date, the Seller Parties shall not, and shall not permit their Affiliates to, solicit or hire any Employee, other than a Non-Continuing Employee; provided, however, nothing herein shall restrict the Seller Parties or their Affiliates from (i) soliciting any such Employee by general employment advertising or third party employment agencies (so long as such agencies are not directed by the soliciting Seller or Affiliate to target such Employees), or (ii) hiring any Employee who responds to such permitted solicitation or seeks employment on an unsolicited basis.
- (f) Except as set forth in <u>Section 6.5(e)</u>, for a period of five years from the Closing Date (i) the Buyers shall indemnify the Sellers for any payments made or Liabilities incurred by the Sellers or any of their Affiliates under Mexican Law attributable to periods after the Closing Date in respect of any of the Mexican Employees and (ii) the Sellers shall indemnify the Buyers and the Mexican Subsidiaries for any payments made or Liabilities (except Liabilities reflected on the Mexican Subsidiaries' balance sheets as of the Closing Date) incurred by the Buyers, their Affiliates and the Mexican Subsidiaries attributable to periods before the Closing Date in respect of any of the Mexican Employees. The amounts specified in this <u>Section 6.5(g)</u> shall be paid and satisfied by the delivery of a certified check or by a wire transfer to the bank account specified by notice to the paying Party.

6.6 Employee Benefit Plans.

(a) Effective as of the Closing Date, all Employees shall cease to participate in the Seller's Employee Benefit Plans. After June 22, 2004, Buyers shall make available

their applicable Employee Benefit Plans to all Continuing Employees as permitted under the terms of such plans.

- (b) As soon as practicable following Closing, the applicable Seller shall take all action necessary: (i) to permit the Employees to elect to take distributions (subject to applicable law) of their accounts under the applicable Seller's defined contribution retirement plan or plans applicable to the Employees (the "Retirement Plan") in accordance with the terms of such Retirement Plan; and (ii) to the extent that the Employees so elect, to roll over the amounts received from the applicable Seller's Retirement Plan (including, to the extent permissible under applicable law, any outstanding loans) to an individual retirement account, or in the case of Continuing Employees, to a defined contribution retirement plan qualified under Section 401(a) of the Code and maintained by Buyer or one of its Affiliates if one is so maintained.
- (c) Nothing contained in this <u>Section 6.6</u>, whether express or implied, is intended to confer upon any Employee or former employee of any of the Seller any right or remedy, including any right as a third-party beneficiary.

6.7 Non-Use of Core Lab Marks.

- (a) The Core Lab Marks appear on some of the properties and assets used in Business, including on signage, and on supplies, materials, stationery, brochures, advertising materials, manuals and similar consumable items used in Business. The Buyers acknowledge and agree that they do not have and, upon consummation of the transactions contemplated by this Agreement, will not have, any right, title, interest, license or other right to use the Core Lab Marks.
- (b) The Buyers will, within three (3) months after the Closing Date, (i) remove the Core Lab Marks from, or cover or conceal the Core Lab Marks on, the Purchased Assets, including signage, (ii) return or destroy all papers and brochures included in the Purchased Assets that contain any Core Lab Marks that are not so covered or removed, and (iii) submit notifications or amendments with respect to any Permits included in the Purchased Assets that use any Core Lab Marks, and Buyers will, during such period, have the limited right to use the Core Lab Marks consistent with this provision and, provided that Buyers' use of the Core Lab Marks will not be reasonably likely to create confusion to the public.
- (c) The Buyers shall not challenge the Seller Parties' or their Affiliates' ownership of the Core Lab Marks or any application for registration thereof or any registration thereof or any rights of the Seller Parties or their Affiliates therein as a result, directly or indirectly, of the Buyers' ownership of the Purchased Assets or operation of Business following the Closing.
- (d) The Buyers will not conduct any business or offer any goods or services under any Core Lab Marks, except as expressly permitted during the transition period pursuant to Section 6.7(b) above or as permitted pursuant to the Seller Parties' written consent.

- (e) The Buyers will not send, or cause to be sent, any correspondence or other materials to any Person on any stationery that contains any Core Lab Marks or otherwise operate Business or the Purchased Assets in any manner that would or might reasonably be expected to confuse any Person into believing that the Buyer has any right, title, interest or license to use any Core Lab Marks, except as expressly permitted during the transaction period pursuant to Section 6.7(b) above or as permitted pursuant to the Seller Parties' written consent.
- 6.8 Insurance Coverage. Following the Closing, the Seller Parties and their Affiliates will have no obligation to provide any insurance coverage necessary, desirable or appropriate for Business or the Purchased Assets. The Seller Parties also, if so requested by any Buyer, will file claims under any of the insurance policies listed on Exhibit 3.18 of the Sellers' Disclosure Schedule that insure any of the Purchased Assets on an "occurrence" basis if the occurrence was before the Closing Date.

6.9 No Competition.

- (a) For the limited and reasonable period of three (3) years following the Closing Date (the "Restricted Period"), none of the Seller Parties shall, or shall permit their Affiliates to, engage in providing seismic data processing, time and depth imaging, seismic signal enhancement or Advanced Reservoir Geophysics services (the "Restricted Business") anywhere in the world, other than as set forth in the License Agreement, provided, however, that nothing herein shall restrict or prevent (i) the Seller Parties or their Affiliates from owning as a passive investor less than ten (10%) percent of any class of securities of any Person that are listed for trading on a recognized stock exchange or national securities exchange, or (ii) any acquisition by Seller Parties or any of their Affiliates of another Person, or the business of another Person, that has an ancillary division, department, subsidiary or other Affiliate, that engages in the Restricted Business (a "Competing Business") but which Person is not principally engaged therein; provided, further, however, that (A) promptly after acquiring such Competing Business, the acquiring Seller Party or such Affiliate shall notify Buyer thereof and (B) if greater than 12 months remain in the Restricted Period, the acquiring Seller Party or such Affiliate shall use Commercially Reasonable Efforts to divest itself of such Competing Business as soon as reasonably practicable.
- (b) The Seller Parties acknowledge that the obligation under Section 6.9(a) forms a significant part of the value for which the Purchase Price is being paid. The Parties stipulate that the obligation under Section 6.9(a) is reasonable and necessary and tailored to Business, which currently is conducted throughout the world from the United States, Canada and Mexico. If nonetheless a court, arbitrator or other Person having jurisdiction concludes in a proceeding that the scope, time or geographic coverage of this Section 6.9 is not enforceable, the Parties agree that the obligation should be revised to a form that is enforceable to the broadest extent permitted by applicable Law.
- (c) For two (2) years following the Closing, the Seller Parties will, and will cause their Affiliates to, refer all inquiries they may receive concerning Business to the

Buyers, and will for one year after the Closing include on any of their web sites that pertain to the Business a pointer to a web site designated in writing by Buyers.

6.10 Collection of Accounts Receivable.

- If the accounts receivable listed on Exhibit 6.10 (the "Specified (a) Receivable"), is not collected in full (net of related reserves) within 360 days after the Closing Date, then notwithstanding any other provision of this Agreement to the contrary, the sole remedy of the Buyers shall be to cause the Seller Parties to repurchase such Specified Receivable at 100% of the net book amount thereof (taking into account any related reserves in effect as of the Closing Date and any payments with respect to such Specified Receivable made after the Closing Date) and no claim may be made based on the inaccuracy of any representation or warranty resulting therefrom. If the Buyers and Seller Parties disagree as to the amount of the Specified Receivable to be repurchased, Buyers and Seller Parties agree to resolve such matter in accordance with Section 9. If the appropriate Buyer elects to cause the Seller Parties to repurchase such Specified Receivable, any notice of such election shall be given in accordance with Section 10.8 and must be received by the Seller Parties prior to 450 days after the Closing Date or the Buyers will be deemed to have waived any rights with respect to such Specified Receivable. Such notice shall (i) identify a closing date not less than 5 days nor more than 90 days after such notice is received by the Sellers upon which the closing of the sale shall occur and (ii) state the price for the repurchase of the Specified Receivable, together with such supporting detail as may be requested by the Sellers (which may be audited by Sellers from time to time upon reasonable advance notice). Such notice shall be accompanied by Buyers' certification of its attempts to collect the applicable Specified Receivable and Buyers shall promptly provide Sellers with all additional records relating to collection attempts with respect to the Specified Receivable as Seller may reasonably request from time to time. At the closing, the appropriate Buyer (or Subsidiary thereof) shall deliver to the purchasing Seller such instruments of transfer as may be reasonably requested by such Seller to evidence the transfer of the Specified Receivable to such Seller, and Seller shall pay the repurchase price in immediately available funds.
- (b) Following the Closing Date, but prior to the repurchase of the Specified Receivable by any Seller, Buyers shall, and shall cause the Subsidiary owning the Specified Receivable to, use good faith efforts to collect the Specified Receivable provided that the Subsidiary is not required to institute litigation or arbitration proceedings or to incur material out of pocket expenses. Any payment received by the Subsidiary or Buyers from the account debtor or the other Person identified in Exhibit 6.10 on account of the Specified Receivable or on account of the specific work giving rise to the Specified Receivable shall be credited to the Specified Receivable.
- (c) If the Specified Receivable is repurchased by any Seller, if requested by such purchasing Seller, Buyer or the relevant Subsidiary of Buyer, shall, to the extent permissible under applicable Law, act as collection agent for such Seller with respect to such Specified Receivable. In such capacity, Buyer or the relevant Subsidiary of Buyer shall (i) diligently pursue collection of such Specified Receivable as if such receivables were owned by Buyer or such Subsidiary provided that the Subsidiary is not required to

30

PATENT REEL: 028590 FRAME: 0039 institute litigation or arbitration or incur material out of pocket expenses, and (ii) remit monies received by Buyer or such Subsidiary in such capacity to the appropriate Seller within 3 Business Days of receipt.

(d) Within sixty (60) days following the Closing Date, the Sellers will provide a work order or, failing that, other identifying support, identifying the basis for the Specified Receivable reasonably satisfactory to Buyer. If such work order or other identifying support is not provided within ninety (90) days, the relevant Subsidiary of Buyers shall have the right to cause the Specified Receivable to be repurchased in accordance with the provisions of Section 6.10(a), and the provisions of Section 6.10(a) and (c) shall then apply.

7. [INTENTIONALLY DELETED.]

8. SURVIVAL; INDEMNIFICATION.

8.1 Survival of Representations, Warranties and Covenants.

- (a) The representations and warranties of the Parties contained in this Agreement will survive following the Closing, subject to the provisions of <u>Sections 8.2(b)</u> and <u>8.3(b)</u>.
- (b) The covenants of the Parties contained in this Agreement will survive the Closing in accordance with their respective terms.

8.2 Indemnification Provisions for Benefit of the Buyers.

- (a) If any Seller Party breaches any of its representations or warranties contained in this Agreement or any of its covenants contained in this Agreement, and subject to the limitations set forth in Sections 8.1(b) and 8.5, then the Seller Parties, jointly and severally, shall indemnify and hold harmless the Buyers and their Representatives (each, a "Buyers' Indemnitee") from and against any Damages any Buyers' Indemnitee may suffer through and after the date of the claim for indemnification resulting from, arising out of, relating to, in the nature of, or caused by such breach.
- (b) To be entitled to indemnification under <u>Section 8.2(a)</u>, a Buyers' Indemnitee must notify the Seller Parties of the alleged breach of the representation or warranty on or before the date occurring:
 - (i) In the case of <u>Section 3.7</u>, three months after the expiration of the applicable statute of limitations;
 - (ii) In the case of <u>Sections 3.10</u> and <u>3.13</u>, five (5) years after the Closing Date; or
 - (iii) In all other cases, eighteen (18) months after the Closing Date.

ASSET PURCHASE AGREEMENT

8.3 Indemnification Provisions for Benefit of the Seller Parties.

- (a) If any Buyer breaches any of its representations or warranties contained in this Agreement or any of its covenants contained in this Agreement, but subject to the limitations set forth in Section 8.3(b), then the Buyers, jointly and severally, shall indemnify and hold harmless the Seller Parties and their Representatives (each, a "Sellers' Indemnitee") from and against any Damages any Sellers' Indemnitee may suffer through and after the date of the claim for indemnification resulting from, arising out of, relating to, in the nature of, or caused by such breach.
- (b) To be entitled to indemnification under <u>Section 8.3(a)</u>, a Sellers' Indemnitee must notify the Buyers of the alleged breach of the representation or warranty on or before the date occurring eighteen (18) months after the Closing Date.

8.4 Matters Involving Third Parties.

- (a) If any third party notifies any Party (the "Indemnified Party") with respect to any matter (a "Third Party Claim") that may give rise to a claim for indemnification against any other Party (the "Indemnifying Party") under this Section 8, then the Indemnified Party will promptly notify each Indemnifying Party in writing. Delay on the part of the Indemnified Party in notifying any Indemnifying Party will not relieve the Indemnifying Party from its obligation unless (and then solely to the extent) the Indemnifying Party is prejudiced thereby.
- (b) Any Indemnifying Party will have the right to defend the Indemnified Party against the Third Party Claim with counsel of the Indemnifying Party's choice reasonably satisfactory to the Indemnified Party so long as (i) the Indemnifying Party notifies the Indemnified Party in writing within 15 days after the Indemnified Party has given notice of the Third Party Claim that the Indemnifying Party will assume defense of such matter, (ii) the Indemnifying Party will have the financial resources to defend against the Third Party Claim and fulfill any indemnification obligations, (iii) the Third Party Claim involves only money damages and does not seek an injunction or other equitable relief, (iv) settlement of, or an adverse judgment with respect to, the Third Party Claim is not, in the good faith judgment of the Indemnified Party, likely to establish a precedential custom or practice materially adverse to the continuing business interests of the Indemnified Party, and (v) the Indemnifying Party conducts the defense of the Third Party Claim actively and diligently.
- (c) So long as the Indemnifying Party is conducting the defense of the Third Party Claim in accordance with Section 8.4(b), (i) the Indemnified Party may retain separate co-counsel at its sole cost and expense and participate in the defense of the Third Party Claim, (ii) the Indemnified Party will not consent to the entry of any judgment or enter into any settlement with respect to the Third Party Claim without the prior written consent of the Indemnifying Party (not to be withheld unreasonably), and (iii) the Indemnifying Party will not consent to the entry of any judgment or enter into any settlement with respect to the Third Party Claim without the prior written consent of the Indemnified Party (not to be withheld unreasonably).

32

PATENT

(d) If any of the conditions in <u>Section 8.4(b)</u> is not or is no longer satisfied, (i) the Indemnified Party may defend against, and consent to the entry of any judgment or enter into any settlement with respect to, the Third Party Claim in any manner it reasonably may deem appropriate (and the Indemnified Party need not consult with, or obtain any consent from, any Indemnifying Party), (ii) the Indemnifying Party will reimburse the Indemnified Party promptly and periodically for the costs of defending against the Third Party Claim (including reasonable attorneys' fees and expenses), and (iii) the Indemnifying Party will remain responsible for any Damages the Indemnified Party may suffer resulting from, arising out of, relating to, in the nature of, or caused by the Third Party Claim to the fullest extent provided in this <u>Section 8</u>.

8.5 Limitations on Sellers' Indemnification Obligations.

- (a) The Seller Parties will have no obligation to indemnify any Buyers' Indemnitee from and against any Damages resulting from, arising out of, relating to, in the nature of or caused by the breach or alleged breach by any Seller Party of its representations or warranties contained in this Agreement until the Buyers' Indemnitees have suffered Damages by reason of all such breaches (or alleged breaches) in excess of \$400,000 as an aggregate deductible (at which point the Seller Parties will be obligated only to indemnify the Buyers' Indemnitees from and against Damages in excess of such deductible); provided, however, that this limitation shall not apply to breaches of the representations and warranties in Section 3.7. In calculating amounts of Damages for purposes of this Section 8.5(a), references to Material Adverse Effect or other limitations on materiality or amount shall be disregarded.
- (b) In the absence of fraud, in no event shall the Seller Parties be obligated to expend in excess of an amount (the "Cap") equal to \$10,500,000, in the aggregate to indemnify the Buyers' Indemnitees from and against all Damages resulting from, arising out of, relating to, in the nature of or caused by the breach or alleged breach by the Seller Parties of their representations or warranties contained in this Agreement or their covenants contained in this Agreement; provided, however, that
 - (i) with respect to any breach or alleged breach of the Seller Parties' representations and warranties or covenants, other than the representations and warranties set forth in Section 3.13, the Cap shall be \$6,300,000, and
 - (ii) with respect to any breach or alleged breach of the Seller Parties' representations and warranties set forth in Section 3.13, the Cap (inclusive of any Damages limited by clause (i)) shall be:
 - (A) \$10,500,000, for any breach or alleged breach of such representations and warranties limitations as to which notice is given by Buyer's Indemnitee (a "Claim Notice") on or before eighteen (18) months after the Closing Date;

- (B) \$9,450,000, for any such breach or alleged breach as to which a Claim Notice is given more than eighteen (18) months, but on or before twenty-four (24) months, after the Closing Date;
- (C) \$8,400,000, for any such breach or alleged breach as to which a Claim Notice is given more than twenty-four (24) months, but on or before thirty-six (36) months, after the Closing Date:
- (D) \$6,300,000, for any such breach or alleged breach as to which a Claim Notice is given more than thirty-six (36) months, but on or before forty-eight (48) months, after the Closing Date; and
- (E) \$4,200,000, for any such breach or alleged breach as to which a Claim Notice is given more than forty-eight (48) months, but on or before sixty (60) months, after the Closing Date.

<u>provided</u>, <u>further</u>, that the limitations of this <u>Section 8.5(b)</u> shall not apply to breaches of the representations and warranties in <u>Sections 3.5(a)</u> and 3.7.

8.6 Other Indemnification Provisions.

- (a) The Parties will make appropriate adjustments for Tax benefits, insurance coverage and third-party indemnification rights in determining Damages for purposes of this Section 8. The Seller Parties shall not be liable for any Damages to the extent that the matter giving rise to such Damages is the subject of an allowance, provision, reserve or adjustment in the finally determined in accordance with Section 2.4. All indemnification payments under this Section 8 will be deemed adjustments to the Purchase Price.
- (b) The Sellers' Indemnitees and the Buyers' Indemnitees, as appropriate, will use Commercially Reasonable Efforts to mitigate all Damages relating to claims for indemnification pursuant to this Section 8, including availing itself of any defenses, limitations, rights of contribution, claims against third parties and other rights at law or equity. For purposes of this Section 8.6(b), Commercially Reasonable Efforts will include the reasonable expenditure of money to mitigate or otherwise reduce or eliminate any loss or expenses for which indemnification would otherwise be due (which reasonable expenditures will be included as Damages suffered by the Party incurring such expenditures).
- (c) Notwithstanding the provisions of <u>Section 8.5</u> or any other provision of this Agreement, the limitations on the indemnification obligations set forth therein shall not apply to (i) the failure of any Seller Party to perform its obligations with respect to the Excluded Liabilities, (ii) any Buyer's failure to perform its obligations with respect to the Assumed Liabilities, and (iii) failures to perform the obligations of the Parties with respect to the prorations in <u>Section 2.9</u>.
- (d) The Parties acknowledge that each of the Sellers' Indemnitees and the Buyers' Indemnitees is an intended third-party beneficiary under this Section 8 and that

34

ASSET PURCHASE AGREEMENT

any of the Sellers' Indemnitees or the Buyers' Indemnitees will have all legal rights, remedies and claims under this <u>Section 8</u>.

8.7 Exclusive Remedy. Except as set forth in Section 8.6(c), the right of the Parties and their respective Representatives to assert claims and receive payments under this Section 8 is the sole and exclusive right and remedy exercisable by the Parties and their respective Representatives with respect to any Damages arising out of any breach by any Party of any representation or warranty of such Party set forth in this Agreement. No Party or its Representatives will have any other remedy (statutory, equitable, common law or otherwise) against any other Party with respect to such breaches, and all such other remedies are hereby waived. Without limiting the foregoing, each of the Parties acknowledges and agrees that neither it nor its Representatives will have any remedy after the Closing for any breach of the representations and warranties contained in this Agreement, except as expressly provided in Sections 8.2 and 8.3 of this Agreement.

8.8 Disclaimer of Other Representations and Warranties.

- (a) EXCEPT FOR THOSE REPRESENTATIONS AND WARRANTIES EXPRESSLY SET FORTH IN SECTION 3, THE SELLER PARTIES MAKE NO REPRESENTATIONS OR WARRANTIES OF ANY KIND OR NATURE, EXPRESS OR IMPLIED, AS TO THE PURCHASED ASSETS OR BUSINESS, OR THE PROSPECTS (FINANCIAL AND OTHERWISE), RISKS AND OTHER INCIDENTS OF BUSINESS OR THE PURCHASED ASSETS.
- (b) WITHOUT LIMITING THE FOREGOING, AND EXCEPT FOR THOSE REPRESENTATIONS AND WARRANTIES EXPRESSLY SET FORTH IN SECTION 3, THE SELLER PARTIES MAKE NO REPRESENTATIONS OR WARRANTIES OF MERCHANTABILITY, USAGE OR SUITABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE WITH RESPECT TO THE PURCHASED ASSETS, OR ANY PART THEREOF, OR AS TO THE WORKMANSHIP THEREOF, OR THE ABSENCE OF ANY DEFECTS THEREIN, WHETHER LATENT OR PATENT, OR COMPLIANCE OF SUCH PROPERTIES OR ASSETS WITH ANY LAWS, INCLUDING ENVIRONMENTAL LAWS, OR AS TO THE CONDITION OF THE PURCHASED ASSETS, OR ANY PART THEREOF, OR AS TO THE ABSENCE OF HAZARDOUS SUBSTANCES OR LIABILITY OR POTENTIAL LIABILITY UNDER ENVIRONMENTAL LAWS WITH RESPECT TO BUSINESS OR THE PURCHASED ASSETS. ANY SUCH OTHER REPRESENTATIONS AND WARRANTIES ARE HEREBY EXPRESSLY DISCLAIMED.
- (c) EXCEPT AS SET FORTH IN <u>SECTION 3</u>, THE PURCHASED ASSETS ARE SOLD "AS IS, WHERE IS" ON THE CLOSING DATE, AND IN THEIR CONDITION ON THE CLOSING DATE "WITH ALL FAULTS."
- (d) WITHOUT LIMITING THE FOREGOING, NO MATERIAL OR INFORMATION PROVIDED BY OR COMMUNICATIONS MADE BY THE SELLER OR ITS REPRESENTATIVES (INCLUDING ANY BROKER OR INVESTMENT BANKER), INCLUDING ANY INFORMATION OR MATERIAL CONTAINED IN THE DUE DILIGENCE MATERIALS OR IN THE INFORMATION MEMORANDUM WILL CAUSE OR CREATE ANY WARRANTY, EXPRESS OR IMPLIED, AS TO THE TITLE, CONDITION, VALUE OR QUALITY OF THE PROPERTIES OR ASSETS OF BUSINESS OR THE PURCHASED ASSETS.

9. DISPUTE RESOLUTION.

All Disputes shall be resolved in accordance with the dispute resolution procedures set forth in this <u>Section 9</u>. Notwithstanding the foregoing, (a) the Parties may at any time seek injunctive or equitable relief from a court of competent jurisdiction, and (b) nothing herein shall prevent a Party from defending or pursuing any claim in a court or other proceeding initiated by a third party.

9.1 Negotiations By Senior Management.

- (a) In the event of a Dispute among the Parties, the Parties will endeavor to reach a satisfactory solution by referring the Dispute to senior management of each of the Disputing Parties. For this purpose, representatives of each Disputing Party shall reduce the main background information and contended issues to a written report, which shall be a joint report if possible, or, if such a joint report cannot be agreed upon, a report shall be prepared by representatives of each Disputing Party, which report or reports will be submitted to senior management of all Disputing Parties.
- (b) After receiving the written report or reports pursuant to Section 9.1(a), senior management of the Disputing Parties will meet as soon as possible, on no less than seven (7) days' written notice, unless specifically agreed otherwise. Such senior management shall examine the arguments given in the written reports as well as any other submissions by the Disputing Parties, and shall, if the Dispute cannot be resolved immediately, agree to convene for further negotiations aimed at resolving the Dispute.
- (c) Should senior management of the Disputing Parties be unable to resolve the Dispute within twenty-one (21) days, any Disputing Party may, by delivery of written notice to the others, submit the Dispute to binding arbitration in accordance with <u>Section 9.2</u>.

9.2 Binding Arbitration.

- (a) If a Dispute remains unresolved after the discussions of senior management of the Disputing Parties in accordance with Section 9.1, or if one or more of the Disputing Parties fail to comply with any of the time periods set forth in Section 9.1, the Parties agree that upon submittal by any Disputing Party, all such Disputes shall be finally settled by binding arbitration, pursuant to the Commercial Arbitration Rules of the American Arbitration Association (the "AAA").
- (b) If the amount in controversy in any Dispute is less than \$2,000,000, the arbitration shall be conducted by one independent arbitrator, appointed by the AAA. Each Disputing Party shall promptly notify the other Disputing Parties and the AAA in writing if the arbitrator appointed by the AAA has any relationship to or affiliation with such Disputing Party (a "Notice of Relationship"), in which event another arbitrator shall be appointed by the AAA promptly thereafter.
- (c) If the amount in controversy in any Dispute is \$2,000,000 or more, the arbitration shall be conducted by three (3) arbitrators who are to be appointed in

ASSET PURCHASE AGREEMENT

accordance with the said rules and who, to the extent possible, shall have experience and expertise in the subject matter involved in the dispute. Unless all Disputing Parties can agree in writing on the arbitrators within ten (10) days after the delivery of the notice referred to in Section 9.2(a), then, within ten (10) days thereafter, the Sellers, on the one hand (which shall be entitled to specify one arbitrator), and the Buyers, on the other hand (which shall be entitled to specify one arbitrator), shall each notify the other in writing of the name of the independent arbitrator chosen by them to participate as a member of a three-member panel of arbitrators. If either the Sellers or the Buyers fail to give the other timely notice of such appointment, then the Disputing Party who timely gave such notice (the "Notifying Disputing Party") may give notice to the Disputing Party who failed to give timely notice (the "Notified Disputing Party") of the Notifying Disputing Party's intent to require that its arbitrator act as the sole arbitrator hereunder. If the Notified Disputing Party shall not have given notice of appointment of its arbitrator within five (5) Business Days following receipt of notice from the Notifying Disputing Party, then the arbitrator selected by the Notifying Disputing Party shall act as the sole arbitrator hereunder. If an arbitrator is timely appointed by each of the Disputing Parties, the two named arbitrators shall select the third member of the arbitration panel within ten (10) days after they have both been appointed, and they shall promptly notify the Disputing Parties thereof. Each Disputing Party shall promptly deliver to the other Disputing Parties and the Disputing Parties' selected arbitrators a Notice of Relationship if the third arbitrator has any relationship to or affiliation with such Disputing Party, in which event another arbitrator shall be selected within ten (10) days after receipt of such Notice of Relationship by the Disputing Parties' selected arbitrators. If the two initially appointed arbitrators cannot timely agree on a third arbitrator, then any Disputing Party may request that the American Arbitration Association select the third arbitrator.

- (d) The arbitration hearing shall be held in such city as may be agreed by the Disputing Parties, <u>provided</u> that if the Disputing Parties do not agree upon any city within thirty (30) days after delivery of the notice referred to in <u>Section 9.2(a)</u>, the arbitration hearing shall be held in New York City. The hearing will be held at a site in such city to be determined by the sole arbitrator, or as agreed to by a majority of the arbitrators, as the case may be, on thirty (30) days' written notice to the Disputing Parties. The arbitration proceedings shall be held in the English language.
- (e) The arbitration hearing shall be concluded within thirty (30) days unless otherwise ordered by the sole arbitrator or a majority of the arbitrators, as the case may be, on compelling grounds, and the award thereon or decision with respect thereto shall be made within ten (10) days after the close of the submission of evidence. Arbitration demanded hereunder by any Disputing Party shall be final and binding on the Parties and may not be appealed except in the case of manifest error or impropriety in the arbitration proceedings. The decision, arbitration order and relief agreed upon in writing by the sole arbitrator, or by any two or more of the arbitrators (in the case of a three-member panel) shall be deemed the arbitrators decision for all purposes hereof. If two or more members of a three-person arbitration panel cannot agree, then the decision of the arbitrator not appointed by any Disputing Party shall control. The references herein to the arbitration panel shall also be deemed to refer to the single arbitrator where a panel is not being used hereunder, and all references to decisions, orders, awards and relief granted by the panel

of arbitrators shall mean the decision, order, award or relief agreed upon in writing by the required number of members of the panel, as indicated.

- (f) The Parties agree that the arbitration panel may render and the Parties shall abide by any interim ruling that the arbitration panel deems necessary or prudent regarding discovery, summary proceedings, or other pre-arbitration matters.
- (g) The decision of the arbitration panel shall be final and binding on all Parties, and judgment on the award of the arbitration panel may be entered by any court having jurisdiction thereof.
- (h) Any costs or other expenses, including reasonable attorneys' fees and costs incurred by the successful Disputing Party, arising out of or occurring because of the arbitration proceedings may be assessed against the unsuccessful Disputing Party, borne equally, or assessed in any manner within the discretion of the arbitration panel and shall be included as part of any order or decision rendered by the arbitration panel. The arbitration panel may also order any Disputing Party who is ordered to pay any other Disputing Party's attorneys' fees and costs to pay interest on such award at a rate not to exceed the Prime Rate from the date of the award until paid. As an initial matter (and until ordered differently by the arbitration panel in connection with an award), each of the Sellers, on the one hand, and the Buyers, on the other hand, shall pay the fees, costs and expenses charged by the arbitrator chosen by it, and, in advance, one-half (1/2) of the fees, costs and expenses charged by the third arbitrator.
- (i) Third parties dealing with any Party shall be entitled to fully rely on any written arbitration order or decision with regard to the matters addressed therein, whether or not such arbitration order or decision has been confirmed or adopted by a court, or incorporated in any order of any court.

10. MISCELLANEOUS.

- separate press releases upon completion of the Closing, announcing the consummation of the sale of the Purchased Assets. Each of such press releases shall be subject to the review and approval by the other Parties hereto. For a period of one year after the Closing, no Party will issue any other press release or make any other public announcement relating to the subject matter of this Agreement without the prior written approval of the other Parties (other than its own Affiliates). Any Party, however, may make any public disclosure it believes in good faith is required by applicable Law or any listing or trading agreement concerning its or its Affiliates publicly-traded securities (in which case the disclosing Party will use Commercially Reasonable Efforts to advise the other Parties (other than its own Affiliates) prior to making the disclosure).
- 10.2 Cooperation on Tax Matters; Retention of Records. The Parties will cooperate fully, as and to the extent reasonably requested by the other Party, in connection with any audit, litigation or other proceeding with respect to Taxes. Such cooperation will include the retention and (upon another Party's request and at such requesting Party's expense) the provision of records and information that are reasonably relevant to any such audit, litigation or other

proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any materials provided hereunder. The Seller Parties agree (i) to retain, and to cause their Affiliates to retain, all books and records with respect to Tax matters pertinent to Business, the Purchased Assets, the Assumed Liabilities and (to the extent not included in the Purchased Assets) the assets and Liabilities of the Mexican Subsidiaries relating to any taxable period beginning before the Closing Date until the expiration of the statute of limitations of the respective taxable periods, and to abide by all record retention agreements entered into with any taxing authority, and (ii) to give the Buyers and their Representatives reasonable notice prior to transferring, destroying or discarding any such books and records and, if a Buyer so requests, the Seller Parties will allow, and will cause their Affiliates to allow, the Buyers to take possession of such books and records.

- 10.3 No Third-Party Beneficiaries. This Agreement will not confer any rights or remedies upon any Person other than the Parties and their respective successors and permitted assigns and, with respect to Section 8 of this Agreement only, their respective Representatives.
- Exhibits), together with the Confidentiality Agreement and the Transaction Documents, embodies the entire agreement and understanding of the Parties with respect to the transactions contemplated by this Agreement and the Transaction Documents, and supersedes all prior agreements and understandings between or among the Parties with respect to such transactions. There are no representations, warranties, covenants or agreements between or among the Parties with respect to the subject matter set forth in the Transaction Documents, other than those expressly set forth in the Transaction Documents. Without limiting the foregoing, the Buyer acknowledges and agrees that there are no representations, warranties, covenants or agreements between or among the Parties with respect to any material made available to the Buyer pursuant to the terms of the Confidentiality Agreement (including the Due Diligence Materials and the Information Memorandum) except those expressly set forth in the Transaction Documents.
- 10.5 Succession and Assignment. This Agreement will be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. No Party may assign either this Agreement or any of its rights, interests or obligations under this Agreement without the prior written approval of the other Party; provided, however, that the Buyers may substitute one or more of their Affiliates as the purchaser of any particular Purchased Asset or as the Person to assume any particular Assumed Liabilities or Material Agreement. No permitted assignment or substitution, however, will release the assigning Party from primary liability under this Agreement.
- 10.6 Counterparts. This Agreement may be executed in one or more counterparts, each of which will be deemed an original but all of which together will constitute one and the same instrument.
- 10.7 Headings. The section headings contained in this Agreement are inserted for convenience only and do not affect in any way the meaning or interpretation of this Agreement.
- 10.8 Notices. All notices, requests, demands, claims and other communications under this Agreement will be in writing. Any notice, request, demand, claim or other communication

under this Agreement will be deemed duly given on receipt by delivery at or facsimile transmission to the following address; provided, however, that any delivery or transmission outside the recipient's normal business hours at that location shall be deemed to have been received at the next opening of its normal business hours:

If to any Seller:

Core Laboratories c/o Core Laboratories LP 6316 Windfern Houston, Texas 77040 Attn: John D. Denson Fax: (713) 328-2152

with a copy, which does not constitute notice, to:

Andrews Kurth LLP 600 Travis, Suite 4200 Houston, TX 77002 Attn: Sherwood O. Jones Fax: (713) 238-7113

If to the Buyer:

Paradigm Geotechnology B.V. Chobham House Christchurch Way Woking, Surrey GU21 1JP United Kingdom Attn: General Counsel Fax: 44 (1483) 758001

with a copy, which does not constitute notice, to:

Skadden, Arps, Slate, Meagher & Flom LLP 1600 Smith Street, Suite 4400 Houston, TX 77002 Attn: John C. Ale

Fax: (713) 655-5200

Except as expressly disclaimed in a particular notice, any notice given to or by any Buyer shall be deemed to have been given to or by all other Buyers, and any notice to or by any Seller Party shall be deemed to have been given to or by all other Seller Parties. The Buyers or the Seller Parties may change the address to which notices, requests, demands, claims and other communications hereunder are to be delivered to them by giving the other Parties notice in the manner set forth in this Agreement.

- 10.9 Governing Law. This Agreement has been negotiated under and will be governed by and construed in accordance with the law of the State of New York without giving effect to the choice or conflict of law provision or rules thereof (other than Section 5-1401 of the New York General Obligations Law).
- 10.10 Amendments and Waivers. No amendment of any provision of this Agreement will be valid unless the same is in writing and signed by the Buyers and the Seller Parties. No waiver by any Party of any provision of this Agreement or any default, misrepresentation or breach of warranty or covenant under this Agreement, whether intentional or not, will be deemed to extend to any prior or subsequent default, misrepresentation or breach of warranty or covenant under this Agreement or affect in any way any rights arising by virtue of any prior or subsequent such default, misrepresentation or breach of warranty or covenant.
- 10.11 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction will not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction.
- 10.12 Expenses. Each Party will bear its own costs and expenses (including legal fees and expenses) incurred in connection with this Agreement and the contemplated transactions; the fees and expenses of Haynes and Boone LLP shall be for the account of the Seller Parties.
- 10.13 Annexes, Exhibits, Schedules and Disclosure Schedules. The Disclosure Schedules and the Annexes, Exhibits and Schedules referred to in this Agreement are incorporated in this Agreement by reference and are made a part of this Agreement. In the event of any conflict between the provisions of this Agreement (exclusive of the Annexes, Exhibits and Schedules to this Agreement) and any Annex, Exhibit or Schedule, the provisions of this Agreement will govern.
- 10.14 Specific Performance. Each of the Parties acknowledges and agrees that Business and the Purchased Assets are unique and valuable assets and that the other Parties (other than its own Affiliates) would be damaged irreparably in the event any of the provisions of this Agreement is not performed in accordance with its specific terms or otherwise is breached. Accordingly, each of the Parties agrees that the other Parties (other than its own Affiliates) is entitled to an injunction or injunctions to prevent breaches of those provisions of this Agreement and to enforce specifically those provisions, in addition to any other remedy to which it may be entitled, at law or in equity.
- 10.15 Waiver of Consequential Damages. IN NO EVENT WILL ANY PARTY OR ITS REPRESENTATIVES BE LIABLE TO ANY OTHER PARTY OR ITS REPRESENTATIVES UNDER THIS AGREEMENT AT ANY TIME FOR PUNITIVE, CONSEQUENTIAL, SPECIAL, OR INDIRECT LOSS OR DAMAGE, INCLUDING LOSS OF PROFIT, LOSS OF REVENUE OR ANY OTHER SPECIAL OR INCIDENTAL DAMAGES, WHETHER IN CONTRACT, TORT (INCLUDING NEGLIGENCE), STRICT LIABILITY OR OTHERWISE; PROVIDED, HOWEVER, THAT FOR PURPOSES OF THIS SENTENCE ANY DAMAGES AWARDED TO A THIRD PARTY SHALL BE DEEMED TO BE ACTUAL DAMAGES EVEN IF

ASSET PURCHASE AGREEMENT

CONSTITUTING ANY OF THE FOREGOING. EACH PARTY HEREBY EXPRESSLY RELEASES THE OTHER PARTIES AND THEIR RESPECTIVE REPRESENTATIVES FROM ALL SUCH DAMAGES.

10.16 Guaranty. Paradigm Geotechnology guarantees the punctual payment and performance of all obligations of the other Buyers under the Transaction Documents. Seller Parent guarantees the punctual payment and performance of all obligations of the Sellers under the Transaction Documents. This guarantee is a guarantee of payment and not of collection, and may be enforced against the guaranteeing Party without the need of first pursuing, joining or obtaining any judgment or award against the Party whose obligations are so guaranteed. This guarantee is enforceable regardless of any unenforceability against or bankruptcy or insolvency of the Party whose obligations are so guaranteed.

10.17 Electronic Signatures.

- Notwithstanding the Electronic Signatures in Global and National Commerce Act (15 U.S.C. Sec. 7001 et seq.), the Uniform Electronic Transactions Act, or any other Law relating to or enabling the creation, execution, delivery, or recordation of any contract or signature by electronic means, and notwithstanding any course of conduct engaged in by the Parties, no Party shall be deemed to have executed this Agreement or any other Transaction Document (including any amendment or other change thereto) unless and until such Party shall have executed this Agreement or such Transaction Document or other document on paper by a handwritten original signature or any other symbol executed or adopted by a Party with current intention to authenticate this Agreement or such Transaction Document or such other document contemplated.
- Delivery of a copy of this Agreement or any other Transaction Document or such other document bearing an original signature by facsimile transmission (whether directly from one facsimile device to another by means of a dial-up connection or whether mediated by the worldwide web), by electronic mail in "portable document format" (".pdf") form, or by any other electronic means intended to preserve the original graphic and pictorial appearance of a document, or by combination of such means, shall have the same effect as physical delivery of the paper document bearing the original signature. "Originally signed" or "original signature" means or refers to a signature that has not been mechanically or electronically reproduced.

[The remainder of this page is blank. Signature pages follow.]

The Parties have executed this Agreement as of the date first above written.

SELLERS:

CORE LABORATORIES CANADA LIMITED, a Canadian corporation

Ву:	se e				
Name	STEVE L	E E			
Title:	INELIDENT C	INE LARGE	24702151	CHNASA	ATO.
COR!	E LABORATOR ership	IES LP, a D	elaware lin	nited	
By: C	ore Laboratories L	LC, its gene	ral partner		
By:					
Name		·*····································			
Title:					
CORI Mexic	E LAB de MEXIC an <i>sociedad anoni</i>	CO S.A. de e ma de capite	C.V., a al variable		
By:	WW9-11-				
					
Title:		1-1-7			
SELL	ER PARENT:				
CORE	LABORATORI	ES N.V., a	Netherlands	3	
limitec	l liability company	<i>;</i>			
By: Co	ore Laboratories I Managing Direct	nternational	B.V.,		
5016		· ·			
Ву:					
Name:			······································		
Title:	Managing Direct	or of Core L	aboratories		
	International B.V				

S-1

ASSET PURCHASE AGREEMENT

The Parties have executed this Agreement as of the date first above written.

SELLERS:

CORE LABORATORIES CANADA LIMITED, a Canadian corporation

Ву:	
Name:	
Title:	

CORE LABORATORIES LP, a Delaware limited partnership

By: Core Laboratories LLC, its general partner

By: \geq		
Name:	JOHN B. DENSON	
Title:	SECRETARY	

CORE LAB de MEXICO S.A. de C.V., a Mexican sociedad anonima de capital variable

By: Name: Title: GENERAL MANAGER

SELLER PARENT:

CORE LABORATORIES N.V., a Netherlands limited liability company

By: Core Laboratories International B.V., its sole Managing Director

By:

Name:

7. Schauter

Title: Managing Director of Core Laboratories

International B.V.

BUYERS:

PARADIGM GEOTECHNOLOGY B.V., a

Netherlands limited liability company

By:
Name:
Shey Bubey
Title:

Attorney In fact

PARADIGM GEOPHYSICAL CANADA, LTD.,

an Alberta, Canada corporation

By:
Name: Shaj Bubil
Title: Marnos in fact

PARADIGM GEOPHYSICAL CORP., a

Delaware corporation

By:
Name:
Shai Butta
Title:

Title:

ANNEX 1

DEFINED TERMS

- "AAA" has the meaning set forth in Section 9.2(a) hereof.
- "Advanced Reservoir Geophysics" means AVO Processing, Seismic Lambda Mu Rho Processing, Geophysical Impedence Inversion, Geophysical True Depth Conversion, Fault and Fracture Analysis through Geophysical Processing, and Pore Pressure Prediction through Geophysical Processing.
- "Affiliate" means, with respect to any Person, any other Person who controls, is controlled by, or is under common control with, such Person. For purposes of this definition, "control", when used with respect to any Person, means the power to direct or cause the direction of the management and policies of such Person, directly or indirectly, whether through the ownership or voting securities, by contract or otherwise.
- "AGI Mexicana" means AGI Mexicana S.A. de C.V., a Mexican sociedad anonima de capital variable.
 - "Agreement" has the meaning set forth in the preamble.
 - "Arenque Contract" has the meaning set forth in Section 6.1(d).
 - "Assigned Agreements" has the meaning set forth in Section 2.1(i).
 - "Assignment and Assumption Agreement" has the meaning set forth in Section 2.5(a).
 - "Assumed Liabilities" has the meaning set forth in Section 2.5(a).
 - "Audit" has the meaning set forth in Section 2.4.
 - "Business" has the meaning set forth in the recitals to this Agreement.
- "Business Day" means any day other than a Friday, Saturday, a Sunday, or a day on which commercial banks generally are closed in New York City or Tel Aviv, Israel.
 - "Buyer" has the meaning set forth in the preamble to this Agreement.
 - "Buyers' Disclosure Schedule" has the meaning set forth in Section 4.
 - "Buyers' Indemnitee" has the meaning set forth in Section 8.2.
- "Buyers' Required Consents" means those consents and notices that the Buyer must obtain from, or make with, any third party in connection with the transactions contemplated by this Agreement, as set forth on Exhibit 1(a).
- "Canadian Employee" means any Employee identified in Section 3.14 of the Sellers' Disclosure Schedule as being employed in Canada.

ANNEX 1 TO PURCHASE AGREEMENT

- "Cap" has the meaning set forth in Section 8.5.
- "CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980, 42 U.S.C. § 9601 et seq.
 - "Claim Notice" has the meaning set forth in Section 8.5.
 - "Closing" has the meaning set forth in Section 2.7.
 - "Closing Date" has the meaning set forth in Section 2.7.
 - "Closing Date Adjustments" has the meaning set forth in Section 2.4(b).
 - "COBRA" means Section 4980B of the Code and Part 6, Title B of Title I of ERISA.
 - "Code" means the Internal Revenue Code of 1986.
- "Coherence Cube Patents" means United States Patent Nos. 5,563,949; 5,838,564; 5,892,732; RE38,229 (reissue of 5,930,730); 6,092,026; and any divisionals, reissues, continuations, continuations-in-part, renewals and extensions and foreign counterparts of the foregoing.
- "Commercially Reasonable Efforts" means efforts that are reasonably within the contemplation of the Parties at the time of entering into this Agreement and that do not require the performing Party to expend funds other than expenditures that are customary and reasonable in transactions of the kind and nature contemplated by this Agreement in order for the performing Party to satisfy its obligations under this Agreement.
 - "Competing Business" has the meaning set forth in Section 6.9(a).
- "Confidentiality Agreement" means that certain confidentiality agreement between Seller Parent and Paradigm Geotechnology, dated March 2, 2004.
 - "Consortium Contracts" has the meaning set forth in Section 6.1.
 - "Continuing Employee" has the meaning set forth in Section 6.5(c).
- "Core Lab Marks" means the names and marks "Core Laboratories," and "Core Lab," together with all derivations and variations thereof, and the Core Laboratories corporate logos and reservoir optimization logos, together with all derivations or variations thereof. Core Lab Marks does not include Sellers' right, title and interest in and to the name "Coherence Cube."
 - "Core Lab Global" has the meaning set forth in Section 2.1(f).
 - "Core Lab Operations" has the meaning set forth in Section 6.1.
- "Damages" means all losses, damages, penalties, fines, costs, reasonable amounts paid in settlement, Liabilities, obligations, Taxes, liens, expenses and fees, including court costs and reasonable attorneys' fees and expenses.

"Disclosure Schedules" means Buyers' Disclosure Schedule and Sellers' Disclosure Schedule.

"Dispute" means any controversy, claim or dispute that arises out of or in connection with any Transaction Document or the construction, interpretation, performance, breach, termination, enforceability or validity of any Transaction Document, whether the same is based on rights, privileges or interests recognized by or based upon statute, contract, agreement (whether written or oral), tort, common law or other Law.

"Disputing Party" means any Party which is a party to a Dispute.

"Due Diligence Materials" means (a) all due diligence materials provided for review or distributed in written or electronic form by any of the Seller Parties or their Representatives to the Buyers or their Representatives, (b) all written, oral or electronic answers to questions provided by any of the Seller Parties or their Representatives to the Buyers or their Representatives, and (c) all materials contained in data rooms or privately-accessible internet sites established for purposes of providing due diligence materials to the Buyers or their Representatives.

"Employee" means an individual who is employed by the Seller Parties or their Affiliates and working principally in the Business immediately prior to the date of this Agreement.

"Employee Benefit Plan" means any: (a) "employee benefit plan" as such term is defined in ERISA § 3(3); (b) nonqualified deferred compensation or retirement plan or arrangement; (c) qualified defined contribution retirement plan or arrangement that is an Employee Pension Benefit Plan; (d) qualified defined benefit retirement plan or arrangement that is an Employee Pension Benefit Plan; or (e) Employee Welfare Benefit Plan or other similar plan, program or arrangement regardless of whether such plan, program or arrangement is covered by ERISA.

"Employee Pension Benefit Plan" has the meaning set forth in ERISA § 3(2).

"Employee Plan" has the meaning set forth in Section 3.14.

"Employee Welfare Benefit Plan" has the meaning set forth in ERISA § 3(1).

"Employment Agreements" has the meaning set forth in Section 3.14.

"Encumbrances" means any mortgages, pledges, claims, liens, security interests, options, warrants, purchase rights, conditional and installment sales agreements, easements, activity and use restrictions and limitations, exceptions, rights-of-way, deed restrictions, defects or imperfections of title, encumbrances, charges and adverse claims of any kind, except (i) preferential rights of purchase and other consents to or restrictions on the sale or transfer of the shares of the Mexican Subsidiaries contained in the articles of incorporation or charters of the Mexican Subsidiaries as between the respective shareholders of those Subsidiaries, and (ii) preemptive rights of purchase under Mexican Laws or said articles of incorporation or charters in favor of the shareholders of the Mexican Subsidiaries.

"Environmental Laws" means all Laws and the common law: (a) concerning public health and safety, worker health and safety, and pollution or protection of the environment or natural resources, including CERCLA, the Resource Conservation and Recovery Act of 1976, 42 U.S.C. § 6901 et seq., the Clean Air Act, 42 U.S.C. § 7401 et seq., the Federal Water Pollution Control Act, 33 U.S.C. § 1251 et seq., the Oil Pollution Act of 1990, 33 U.S.C. § 2701 et seq., and the Occupational, Safety and Health Act, 29 U.S.C. § 651 et seq., each as amended as of the date of this Agreement; (b) Laws concerning Hazardous Substances; and (c) Laws relating to the management or use of natural resources.

"ERISA" means the Employee Retirement Income Security Act of 1974.

"Excluded Assets" has the meaning set forth in Section 2.2.

"Excluded Liabilities" has the meaning set forth in Section 2.4(b).

"GAAP" means United States generally accepted accounting principles, consistently applied, as in effect from time to time.

"Governmental Authority" means any federal, state, local or foreign governmental, administrative or regulatory authority, court, agency or body, or any division or subdivision, or any arbitrator, arbitration board, tribunal or mediator.

"Governmental Permits" has the meaning set forth in Section 2.1(g).

"Hazardous Substances" means (a) any petrochemical or petroleum products, radioactive materials, radon gas, asbestos in any form, urea formaldehyde foam insulation and transformers or other equipment that contain dielectric fluid that may contain polychlorinated biphenyls, (b) any chemicals, materials or substances, whether solids, liquids, semi-liquids or gas, defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "hazardous constituents," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "contaminants," "pollutants," "irritants," "toxic pollutants" or words of similar meaning and regulatory effect under any applicable Environmental Law, and (c) any other chemical, material or substance, exposure to which is prohibited, limited or regulated by any applicable Environmental Law.

"Income Tax" means any federal, state, local or foreign income Tax or other Tax based upon or related to income or receipts, including any interest, penalty or addition, whether disputed or not.

"Indemnified Party" has the meaning set forth in Section 8.4.

"Indemnifying Party" has the meaning set forth in Section 8.4.

"Information Memorandum" means that Information Memorandum regarding the Business delivered to Buyer in March 2004.

"Intellectual Property" has the meaning set forth in Section 2.1(f).

4

REEL: 028590 FRAME: 0058

"Inventories" has the meaning set forth in Section 2.1(b).

"Knowledge" means the actual knowledge, as of the date of this Agreement or, with respect to any certificate delivered pursuant to this Agreement, the date of delivery of the certificate, of the current directors and executive officers of any of the Seller Parties or the Buyers, as applicable.

"Law" means any federal, state, local or foreign constitution, law, code, plan, statute, rule, regulation, ordinance, order, judgment or decree of any Governmental Authority.

"Leased Real Property" has the meaning set forth in Section 2.1(d).

"Liability" means any liability or obligation of whatever kind or nature, whether known or unknown, whether asserted or unasserted, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated, and whether due or to become due.

"Licenses" has the meaning set forth in Section 2.1(h).

"March 31, 2004 Net Assets Statement" has the meaning set forth in Section 2.4.

"Material Adverse Change" means:

- (a) Cancellation or termination of contracts of Business that in aggregate account for more than \$1,000,000 on an annual basis;
- (b) Amounts unpaid for more than 360 days (whether disputed or otherwise) on Accounts Receivable, and on accounts receivable of AGI Mexicana, in the aggregate exceeding \$1,000,000;
- (c) Any material portion of the Purchased Assets is condemned or taken by eminent domain, or is materially damaged or destroyed by other casualty; or
 - (d) Any other event or circumstance having a Material Adverse Effect.

"Material Adverse Effect" means any change or effect that is materially adverse to Business, operation, properties, financial conditions, assets or Liabilities (including contingent Liabilities) of the Purchased Assets, the Assumed Liabilities and the Mexican Subsidiaries and their assets and Liabilities taken as a whole. "Material Adverse Effect," however, does not include any effect that is attributable to any of the following:

- (e) Any change (or changes taken together) or effect generally affecting the international, national or regional seismic data processing industry as a whole;
- (f) Any change (or changes taken together) or effect resulting from changes in the international, national or regional markets for seismic data processing services;
- (g) Any change (or changes taken together) or effect resulting from changes in the general national or regional economic or financial conditions;

(h) Any change in Law, or any judgments, orders or decrees that apply generally to Persons regardless of whether they participate in businesses similar to Business or have assets or Liabilities similar to the Purchased Assets, the Assumed Liabilities or the assets or Liabilities of the Mexican Subsidiaries.

Any determination as to whether any condition or other matter has a Material Adverse Effect will be made only after taking into account all effective insurance coverages and effective indemnifications with respect to such condition or matter.

- "Material Agreements" has the meaning set forth in Section 3.16.
- "Mexican Agreements" has the meaning set forth in Section 3.13(c).
- "Mexican Employee" means any Employee identified in Section 3.14 of the Sellers' Disclosure Schedule as being employed by a Mexican Subsidiary.
 - "Mexican Subsidiaries" means AGI Mexicana and ST.
 - "Nonassignable Item" has the meaning set forth in Section 2.10.
 - "Non-Continuing Employee" has the meaning set forth in Section 6.5(d).
- "Non-Mexican Business" means the portion of the Business whose operations are not conducted by the Mexican Subsidiaries.
 - "Notified Disputing Party" has the meaning set forth in Section 9.2(b).
 - "Notifying Disputing Party" has the meaning set forth in Section 9.2(b).
- "Ordinary Course of Business" means the ordinary course of business consistent with (a) past custom and practice (including with respect to quantity and frequency), and (b) the requirements of any agreement or any Permit.
 - "Other Receivables" has the meaning set forth in Section 6.1(c).
 - "Paradigm Geotechnology" has the meaning set forth in the preamble to this Agreement.
 - "Party" has the meaning set forth in the preamble to this Agreement.
- "Permitted Encumbrances" means: (a) mechanic's, materialmen's and similar liens and rights arising or incurred in the Ordinary Course of Business securing amounts not yet due or payable; (b) liens for Taxes or other governmental charges or assessments not yet due or payable; (c) purchase money liens and liens securing rental payments under capital lease arrangements; (d) all matters which would be set forth as exceptions in a standard form owner's or leasehold policy of title insurance relating to any Leased Real Property; and (e) zoning, entitlement, conservation restriction and other land use and environmental regulations of any Governmental Authority, but only to the extent there is no existing violation relating to the

Business, the Purchased Assets, the Assumed Liabilities or the Mexican Subsidiaries or their assets or Liabilities.

- "Person" means an individual, a partnership, a corporation, a limited liability company, an association, an entity, a joint stock company, a trust, a joint venture, an unincorporated organization, any other business entity or any Governmental Authority.
- "Prime Rate" means for any day, the "Prime Rate" as published from time to time in the Money Rates section of <u>The Wall Street Journal</u> (U.S. Edition), as such Prime Rate may change from time to time.
- "Property Tax" means any Tax resulting from and relating to the assessment of real or personal property by any Governmental Authority.
 - "Purchase Price" has the meaning set forth in Section 2.3.
 - "PwC" has the meaning set forth in Section 2.4.
- "Release" means any release, spill, leak, discharge, disposal of, pumping, pouring, emitting, emptying, injecting, leaching, dumping or allowing to escape into or through the environment.
 - "Report" has the meaning set forth in Section 2.4.
- "Representatives" of a Person means, collectively, such Person's Affiliates and his, her or its and their respective directors, officers, partners, members, employees, representatives, agents, advisors (including accountants, legal counsel, environmental consultants, engineering consultants and financial advisors), parent entities and other controlling Persons.
 - "Restricted Business" has the meaning set forth in Section 6.9(a).
 - "Restricted Period" has the meaning set forth in Section 6.9(a).
 - "Retirement Plan" has the meaning set forth in Section 6.6.
 - "Seller" has the meaning set forth in the preamble to this Agreement.
 - "Seller Parent" has the meaning set forth in the preamble to this Agreement.
 - "Seller Party" has the meaning set forth in the preamble to this Agreement.
 - "Sellers' Disclosure Schedule" has the meaning set forth in Section 3.
 - "Sellers' Indemnitee" has the meaning set forth in Section 8.3.
- "Sellers' Required Consents" means those consents and notices that the Seller must obtain from, or make with, any third party in connection with the transactions contemplated by this Agreement, as set forth on Exhibit 1(b).

- "Shares" has the meaning set forth in Section 2.1(j).
- "ST" means Servicios Tecnologicos de Yacimientos, S.A. de C.V., a Mexican sociedad anonima de capital variable.
 - "Sublease Agreement" has the meaning set forth in Section 2.8.
- "Subsidiary" means, with respect to any Person, any corporation, partnership, limited liability company or other entity in which any Person (a) has direct or indirect equity or other ownership interest that represents 50% or more of the aggregate equity or other ownership interest in such entity, or (b) has the power to vote or direct the voting of sufficient securities to elect a majority of the directors, governors or persons holding similar positions of such entity.
- "Tax" means any federal, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, customs duties, utility, production, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated, or other tax of any kind whatsoever, including any interest, penalty or addition, whether disputed or not.
- "Tax Return" means any return, declaration, report, claim for refund, or information return or statement relating to Taxes, including any schedule, attachment or amendment.
 - "Third Party Claim" has the meaning set forth in Section 8.4.
- "Transaction Documents" means this Agreement and all other documents and agreements to be entered into by one or more of the Parties in connection with the transactions contemplated by this Agreement.
- "U.S. Employee" means any Employee identified in Section 3.14 of the Sellers' Disclosure Schedule as being employed in the United States.

ANNEX 2

ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS ASSIGNMENT AND ASSUMPTION AGREEMENT (this "Agreement"), is made and executed this 22nd day of April, 2004, by and among CORE LABORATORIES CANADA LIMITED, an Alberta, Canada, corporation ("Core Lab Canada"), CORE LABORATORIES LP, a Delaware limited partnership ("Core Lab LP"), CORE LABORATORIES INTERNATIONAL B.V. ("CLIBV"), and CORE LAB de MEXICO S.A. de C.V., a Mexican sociedad anonima de capital variable ("Core Lab Mexico", and collectively with Core Lab Canada, CLIBV and Core Lab LP, the "Assignors"), and PARADIGM GEOTECHNOLOGY B.V., a Netherlands limited liability company, PARADIGM GEOPHYSICAL CANADA LTD., an Alberta, Canada corporation, PARADIGM GEOPHYSICAL CORP., a Delaware corporation, and PARADIGM GEOPHYSICAL, B.V., a Netherlands limited liability company (collectively, the "Assignees").

ARTICLE I RECITALS

WHEREAS, Core Lab Canada, Core Lab LP and Core Lab Mexico, as Sellers, Core Laboratories N.V., a Netherlands public limited liability company ("Seller Parent") and Assignees are parties to that Asset Purchase Agreement, dated as of April 22, 2004, (the "Purchase Agreement"), which provides for the sale by the Assignors and the purchase by the Assignees of the Purchased Assets (as defined in the Purchase Agreement), and the assumption by the Assignees of the Assumed Liabilities (as defined in the Purchase Agreement); and

WHEREAS, pursuant to the Purchase Agreement, the Seller Parent has agreed to cause the Assignors to assign all of the Purchased Assets to the Assignees; and

WHEREAS, the Assignors are wholly-owned subsidiaries of the Seller Parent and the Assignor and the Seller Parent will realize benefits from the transaction provided for in the Purchase Agreement and the assignments herein provided for; and

WHEREAS, the Assignees desire to assume the Assumed Liabilities;

NOW THEREFORE, for and in consideration of the foregoing, the mutual covenants and agreements hereinafter set forth, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Assignors and Assignee hereby agree as follows:

ARTICLE II DEFINITIONS

2.1 <u>Definitions</u>. Unless otherwise defined herein, all capitalized terms used herein shall have the definitions given such terms in the Purchase Agreement.

ANNEX 2 TO PURCHASE AGREEMENT

ARTICLE III TRANSFER OF THE PURCHASED ASSETS

- 3.1 <u>Transfer and Assignment</u>. Effective as of the date hereof, each Assignor hereby conveys, transfers, assigns and delivers unto the Assignees, as described on <u>Attachment I</u> to this Agreement, their successors and assigns, all of such Assignor's right, title and interest in, to and under the Purchased Assets,
- TO HAVE AND TO HOLD the Purchased Assets unto the Assignees, and their successors and assigns forever, and each Assignor does hereby bind itself and its successors and assigns, to WARRANT and FOREVER DEFEND, all and singular the Purchased Assets unto the Assignee and their successors and assigns, against every person whomsoever lawfully claiming or to claim the same, or any part thereof by, through and under such Assignor but not otherwise.
- 3.2 <u>Assumption of Assumed Liabilities</u>. Effective as of the date hereof, each Assignee hereby assumes and agrees to pay, perform and discharge the Assumed Liabilities allocated to that Assignee as described on <u>Attachment I</u> to this Agreement.
- 3.3 Purchased Assets Conveyed As Is, Where Is. EACH ASSIGNEE HEREBY ACKNOWLEDGES AND AGREES THAT, EXCEPT FOR ANY REPRESENTATIONS AND WARRANTIES EXPRESSLY SET FORTH HEREIN OR IN THE PURCHASE AGREEMENT, NO WARRANTIES OR REPRESENTATIONS OF ANY KIND OR CHARACTER, EITHER EXPRESS OR IMPLIED, ARE MADE BY THE ASSIGNORS OR ARE RELIED UPON BY THE ASSIGNEE WITH RESPECT TO THE PURCHASED ASSETS, OR ANY PORTION THEREOF, INCLUDING, WITHOUT LIMITATION, (a) ANY IMPLIED OR EXPRESS WARRANTY OF MERCHANTABILITY, (b) ANY IMPLIED OR EXPRESS WARRANTY OF FITNESS FOR A PARTICULAR PURPOSE, (c) ANY IMPLIED OR EXPRESS WARRANTY OF CONFORMITY TO MODELS OR SAMPLES OF MATERIALS, (d) ANY RIGHTS OF THE ASSIGNEE UNDER APPROPRIATE STATUTES TO CLAIM DIMINUTION OF CONSIDERATION, AND (e) ANY CLAIM BY THE ASSIGNEE FOR DAMAGES BECAUSE OF DEFECTS, WHETHER KNOWN OR UNKNOWN, WITH RESPECT TO THE ASSIGNED INTERESTS, IT BEING THE EXPRESS INTENTION OF THE ASSIGNORS AND THE ASSIGNEES THAT, EXCEPT AS EXPRESSLY SET FORTH HEREIN OR IN THE PURCHASE AGREEMENT, THE PURCHASED ASSETS SHALL BE CONVEYED AND TRANSFERRED TO THE ASSIGNEES IN THEIR PRESENT CONDITION AND STATE OF REPAIR, "AS IS" AND "WHERE IS", WITH ALL FAULTS.

ARTICLE IV RELEASE

Each of the Assignors, for itself and its Affiliates, hereby releases each of AGI Mexicana, S.A. de C.V., ("AGI") and Servicios Technologicos de Yacimientos S.A. de C.V. (formerly known as Core Laboratories Management S.A. de C.V.) ("SI") from all Liabilities to any of the assignors and their Affiliates, except for Liabilities of ST as set forth in that Employer Substitution Agreement dated as of April 20, 2004, between ST and Owen de Mexico, S.A. de

2

ANNEX 2 TO ASSET PURCHASE AGREEMENT C.V. and any Liabilities of AGI or ST to the Seller Parties or their Affiliates arising under the Purchase Agreement.

ARTICLE V MISCELLANEOUS

- 5.1 <u>Successors and Assigns</u>. This Agreement shall bind and inure to the benefit of the parties hereto and their respective successors and assigns.
- 5.2 <u>Titles and Captions</u>. All article or section titles or captions in this Agreement are for convenience only, shall not be deemed part of this Agreement and in no way define, limit, extend or describe the scope or intent of any provisions hereof.
- 5.3 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, and each counterpart hereof shall be deemed to be an original instrument, but all such counterparts shall constitute but one instrument.
- 5.4 <u>Further Assurances</u>. The parties agree to take all such further actions and execute, acknowledge and deliver all such further documents that are necessary or useful in carrying out the purposes of this Agreement. Without limitation of the foregoing, Assignor agrees to execute, acknowledge and deliver to the Assignee all such other additional instruments, notices, and other documents and to do all to more fully and effectively grant, convey and assign to the Assignee the Assigned Interests conveyed hereby and intended so to be.
- 5.5 <u>Assignors' Liabilities Several</u>. The liability of each of the Assignors hereunder shall be several and not joint.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first above written.

ASSIGNORS:

CORE LABORATORIES CANADA LIMITED, a Canadian corporation By: Name: Title: CORE LABORATORIES LP, a Delaware limited partnership By: Core Laboratories LLC, its general partner By: Name: Title: CORE LAB de MEXICO S.A. de C.V., a Mexican sociedad anonima de capital variable By: Name: CORE LABORATORIES INTERNATIONAL B.V., a Netherlands limited liability company

Name: ______

By:

ASSIGNEES:
PARADIGM GEOTECHNOLOGY B.V., a Netherlands limited liability company
By: Name: Title:
PARADIGM GEOPHYSICAL CANADA LTD. an Alberta, Canada corporation
By: Name: Title:
PARADIGM GEOPHYSICAL CORP., a Delaware corporation
By: Name: Title:

Attachment I

PARADIGM GEOTECHNOLOGY B.V.

All shares of capital stock of the Mexican Subsidiaries, as owned and assigned by, Core Lab Mexico.

PARADIGM GEOPHYSICAL B.V.

One share of each of the Mexican Subsidiaries as owned, and assigned by, CLIBV.

PARADIGM GEOPHYSICAL CORP.

All Purchased Assets physically located in, and all contracts, agreements and Assumed Liabilities principally performed in or sourced from, the United States, and all other Assumed Liabilities not otherwise allocated hereunder.

PARADIGM GEOPHYSICAL CANADA LTD.

All Purchased Assets physically located in, and all contracts, agreements and Assumed Liabilities principally performed in or sourced from, Canada.

6

ANNEX 2 TO ASSET PURCHASE AGREEMENT

PATENT REEL: 028590 FRAME: 0068

ANNEX 3

LICENSE AGREEMENT

[See attached License Agreement]

ANNEX 3 TO PURCHASE AGREEMENT

PATENT LICENSE AGREEMENT

This Patent License Agreement ("Agreement") is effective as of April 22, 2004, by and between CORE LABORATORIES GLOBAL NV, a Netherlands Antilles corporation ("Licensor"), having its offices at Plasa Smeets 6, Willemstad, Curacao, Netherlands Antilles, and PARADIGM GEOTECHNOLOGY BV, a Netherlands limited liability company ("Licensee"), with offices at Chobham House, Christchurch Way, Woking Surrey GU21 1JP, United Kingdom.

WHEREAS, Licensor is the assignee of the patents listed in Exhibit 1 attached hereto; and

WHEREAS, Licensee desires to obtain a license under the patents and patent applications identified in Exhibit 1 together with any continuations, continuations-in-part and divisional applications thereof and all patents issuing thereon (including reissues, renewals, reexaminations and foreign counterparts of the foregoing) (all of the foregoing, collectively, the "Patents"), and Licensor desires to grant such a license in the Patents to Licensee, subject to the terms, conditions, rights, restrictions and obligations of this Agreement.

NOW, THEREFORE, in consideration of the promises and agreements set forth herein, and with intent to be legally bound hereby, the parties agree as follows:

1. Definitions.

- 1.1 "Licensed Methods and Apparatuses" means any and all methods and apparatuses, including without limitation Coherence Cube® software methods and apparatuses, covered by the claims of the Patents.
- 1.2 "Commercial Sublicensing" means any and all sublicensing permitted by the Grant in Section 2.1 hereof, including without limitation the sublicensing of Licensed Methods and Apparatuses, including without limitation the sublicensing of Coherence Cube® software.
- 1.3 "Commercial Use" means any and all use permitted by the Grant in Section 2.1 hereof, including without limitation the use of Licensed Methods and Apparatuses, including without limitation the processing and display of seismic data using Coherence Cube® software.
- 1.4 "Earned Royalties" shall mean the royalties payable by Licensee to Licensor as set out in Section 3.2 and Section 4 of this Agreement.
- 1.5 "Initial Fee" shall have the meaning as set forth in Section 3.1 of this Agreement.
- 1.6 "Net Revenues" shall mean the gross revenues received by Licensee from the Commercial Sublicensing and/or Commercial Use, less returns to the extent credit is given or paid for returns. The foregoing deduction from the gross revenues shall be determined in accordance with the generally accepted accounting practices used by Licensee in the ordinary course of business. It is the intent of the parties that Net Revenues shall be based on arms-length

PATENT LICENSE AGREEMENT-Page 1

PATENT REEL: 028590 FRAME: 0070 sales transactions by Licensee to independent third parties and not based on sales between or among Licensee and its affiliates. In the case of any transaction other than in an arm's length transaction exclusively for money, Net Revenues shall be calculated on the value of the consideration received or the fair market price (if higher) in the country of sale or disposal. If Licensee derives revenues from providing goods or services covered by the Patents together with other goods or services not covered by the Patents, only the revenue properly attributable to the former shall count as Net Revenues.

- 1.7 "Patents" shall have the meaning as set forth in the recitals above.
- 1.8 "Affiliate" means, with respect to a party, any entity that controls, is controlled by, or is under common control with such party.

2. License Grant and Limitations.

2.1 Grant of License. Licensor hereby grants to Licensee and its Affiliates a perpetual non-exclusive, non-transferable (except pursuant to Section 23) license under the Patents to (1) make, have made, sell, import and use the methods and apparatuses covered by the subsisting claims of the Patents, and (2) to sublicense to end-users to whom Licensee or its Affiliates have sold (leased or licensed) such apparatus, the right to use such apparatuses under the subsisting claims of the Patents, but all such sublicenses shall be limited to the end-users' use of the Licensed Methods and Apparatuses and such sublicenses shall not permit the end-users or any other entity to make or have made the Licensed Methods and Apparatuses. The foregoing shall not prevent end user sublicensees of any software from installing the software onto servers, or from making customary back-up copies for crash recovery purposes.

2.2 Limitations on Grant.

- 2.2.1 <u>No Implied Licenses.</u> There are no implied licenses under this Agreement.
- 2.2.2 No Right to Grant Sublicenses. Except as expressly granted in Section 2.1 hereof, this license does not include the right to sublicense.
- 2.3 Retention of Rights. Licensor retains all rights not expressly granted by this Agreement; provided that Licensor agrees not to use the Patents in a manner that would interfere with Licensee's full enjoyment of its rights under this Agreement. Except as expressly provided, nothing in this Agreement shall grant any rights, directly or indirectly, to any other party, other than Licensee.
- 2.4 Coherence Cube Software. For the avoidance of doubt, it is understood that:
 (a) the rights granted under this Agreement also include the unrestricted right to freely exploit any Coherence Cube® software covered by the Patents consistent with the terms of this Agreement; and (b) if, due to expiration, abandonment, invalidity and/or unenforceability of the Patents, no claim is subsisting which covers such software, Licensee shall be entitled to exploit such software without the payment of any further consideration to Licensor. Within one (1)

PATENT LICENSE AGREEMENT-Page 2

PATENT REEL: 028590 FRAME: 0071 week of the Effective Date, Licensor shall deliver to Licensee a copy of all source code for such software, together with any other ancillary materials (e.g., documentation, databases, libraries, or other code) necessary to use, develop, and maintain such software. Licensor and Licensee shall maintain in confidence, and shall not use for any purpose outside this Agreement, all non-public information in such source code.

3. Consideration.

In consideration for the license granted hereunder, Licensee agrees to pay to Licensor, as follows:

- 3.1 Initial Payment. An initial lump sum payment (the "Initial Payment") of two-hundred and fifty-thousand United States Dollars (\$250,000.00) shall be due within three (3) days of the signing date of this Agreement. The Initial Payment shall be considered a nonrefundable advance against Earned Royalties, and no further payment of Earned Royalties shall be due except to the extent amount of Earned Royalties exceeds the Initial Payment.
- **3.2** Earned Royalties. Earned Royalties shall be as follows: Earned Royalties shall be equal to ten percent (10%) of the Net Revenues accrued on or before the third (3rd) anniversary of the date of this Agreement, but not to exceed one million United States Dollars (\$1,000,000.00) in the aggregate. At such time as Earned Royalties attributable to periods before the third (3rd) anniversary of the date of this Agreement or totaling one million United States Dollars (\$1,000,000.00) have been paid, the license under this Agreement shall be deemed fully paid-up and non-terminable for any reason whatsoever (including termination of this Agreement). Earned Royalties shall accrue upon Licensee's receipt of money or other valuable consideration from each Commercial Use and/or Commercial Licensing.
- 3.3 Offset for Licensor Revenues. Upon fifteen (15) days after the end of each calendar quarter during the term of this Agreement in which Licensor receives revenues from granting licenses or forbearances to third parties under the Patents other than in settlement of litigation or potential litigation under Section 7 ("Licensor's Non-Litigation Licensing Revenue"), Licensor shall furnish to Licensee a written report setting forth Licensor's Non-Litigation Licensing Revenue for licenses commencing after the Effective Date of this Agreement shall be regarded as a credit against Licensee's Earned Royalties payment obligations, as well as against the \$1,000,000 cap set forth in Section 3.2. Thereafter, such Non-Litigation Licensing Revenue received by Licensor for licenses commencing after the Effective Date of this Agreement shall be promptly remitted to Licensee.

4 Payment.

4.1 Payment. All Earned Royalty payments shall be paid within thirty (30) days after the end of each calendar quarter during which Licensee received revenues or other valuable consideration from each such Commercial Use and/or Commercial Sublicensing. Licensee shall pay the Earned Royalties to Licensor in immediately available funds.

PATENT LICENSE AGREEMENT-Page 3

- 4.2 Calendar Quarter Report. Within thirty (30) days after the end of each calendar quarter following the date of the first accrual of Earned Royalties during the term of this Agreement, Licensee shall furnish to Licensor a written report setting forth the aggregate Net Revenues during such calendar quarter and clearly describing how the Earned Royalties payable thereon were calculated.
- 4.3 Conversion Rate. Earned Royalties shall be payable in U.S. dollars. In the event that conversion of the Net Revenues from a foreign currency is necessary, the exchange rate used shall be the rate in effect for the purchase of United States dollars with such currency as of the last business day of the calendar quarter during which such Net Revenues were received, as published in the Wall Street Journal.
- 4.4 Reporting in the Event of Termination. In the event that this Agreement is terminated at any time other than at the end of a calendar quarter, an Earned Royalties report and payment shall be made within thirty (30) days following the termination of this Agreement.
- 4.5 Taxes Withheld. If any Earned Royalties are subject to withholding or similar tax, Licensee shall pay such additional amounts as shall be required to ensure that the net amounts received by Licensor hereunder (including any applicable tax credits used by Licensor) will equal the full Earned Royalties amounts due under Section 3 hereof. The parties shall do all such lawful acts and things and sign all such lawful deeds and documents as either party may reasonably request of the other party to enable Licensee to take advantage of any applicable legal provision or any double taxation treaties which may prevent or limit any tax being imposed or withheld by taxing authorities with respect to Earned Royalties payable by Licensee hereunder.

5. Books and Records.

- 5.1 Obligation to Keep Books and Records. Licensee must keep accurate books and records of the receipt of Net Revenues. Licensee must preserve these books and records for at least three (3) years from the date of the Earned Royalties to which they pertain.
- 5.2 Right of Inspection. Licensor's representatives or agents are entitled to inspect the books and records described in Section 5.1 upon reasonable notice. Such inspection shall occur no more than twice annually, and shall be conducted during Licensee's normal business hours at Licensee's facilities. Licensor will pay the fees and expenses of these inspections. If an under-reporting of more than ten percent (10%) of the Earned Royalties for the period subject to inspection is discovered, then Licensee will pay the reasonable fees and expenses of that inspection. Licensee shall promptly pay to Licensor the amount of any underpayment discovered.
- 6. Patent Marking. As reasonably requested by Licensor, Licensee shall place all appropriate patent and intellectual property notices, markings, and indicia on the apparatuses made, used or sublicensed within the scope of this Agreement, taking into due consideration any physical space constraints on such marking.

7. Patent Infringement.

PATENT LICENSE AGREEMENT-Page 4

- 7.1 Actions by Licensor. Licensor shall have the exclusive right, but not the obligation, to take appropriate legal action in connection with the actual or any potential infringement with respect to which Licensor has performed a pre-litigation investigation sufficient under Rule 11 of the Federal Rules of Civil Procedure. In connection with all such litigation or potential litigation, Licensor shall consult with Licensee and give due deference to the suggestions and preferences of Licensee. Licensee shall use its reasonable efforts to cooperate with and assist Licensor in any action (including litigation) pertaining to the Patents by providing documentary evidence, employee testimony or the like. In the event Licensor shall elect to take such action, the conduct of the action shall be entirely directed and controlled by Licensor, and Licensor shall receive all recoveries from such action, including benefits from received from such actual or potential infringers who settle prior to or after litigation is instituted. To assist in the discovery of third party infringement it is agreed that upon discovery by Licensee of any infringement of the Patents, or the possibility thereof, Licensee shall promptly notify Licensor.
- Requests by Licensee. In the event that Licensor determines it does not wish to 7.2 take such action against a potential or actual infringer, if Licensee decides that it wishes for Licensor to initiate patent infringement litigation against an actual or potential infringer, Licensor shall do so if Licensor has concluded that such infringement litigation complies with the requirements of Rule 11 of the Federal Rules of Civil Procedure and the other requirements of the patent law. In such case, Licensee shall control the litigation with counsel of its choosing and shall pay all out-of-pocket costs and expenses of such litigation, including all costs incurred in connection with any counterclaims that may be lodged by the infringer (and any amounts expended by Licensee therefor (including patent attorney's fees) shall be deemed to reduce Licensee's Net Revenues by an equal amount, for purposes of Licensee's Earned Royalties payment obligations), and Licensee shall receive all recoveries from such action, including benefits received from such actual or potential infringers who settle prior to or after litigation is instituted. Licensor shall have the right to retain counsel of its own choosing to represent its interests in connection with any such litigation. Licensee shall hold harmless and fully indemnify Licensor from and against any and all costs and expenses (including reasonable attorneys fees), liabilities and judgments that may be sustained by Licensor as a result of such claims or counterclaims.

8. Ownership and Prosecution of the Patents.

- 8.1 Licensee acknowledges that Licensor is the sole owner of the Patents.
- 8.2 Licensee further agrees that it does not obtain any interest in the Patents except for the rights granted herein. Licensee agrees not to take any action challenging or opposing, on any grounds whatsoever, the ownership by Licensor of the Patents. Furthermore, Licensee agrees not to contest the validity or enforceability, nor assist or request any third party to contest the validity or enforceability of any of the Patents.
- 8.3 Maintenance of the patents (excluding patent infringement actions). Licensor shall consult with Licensee with respect to the prosecution and maintenance of the Patents in the United States Patent Office or other foreign patent offices, as well as defending against any

PATENT LICENSE AGREEMENT-Page 5

challenges in the United States Patent Office or other foreign patent offices to the scope, validity, and enforceability of the Patents. Unless Licensor has first consulted with Licensee and offered Licensee the option to conduct such prosecution or maintenance, Licensor shall use its best commercially reasonable efforts to continue to prosecute and maintain the Patents, including defending against any challenges in the United States Patent Office or other foreign patent offices to the scope, validity, and enforceability of the Patents. Without limiting the generality of the foregoing, Licensor shall not abandon or permit a lapse of any of the Patents without first offering to Licensee the right to assume responsibility for continued prosecution and/or maintenance thereof, at Licensee's sole discretion and expense. However, if Licensor decides not to prosecute, maintain, or defend a Patent or Patents in the United State Patent Office or any foreign patent office and after consultation Licensee elects not to assume responsibility for the prosecution or maintenance of the Patent or Patents in issues, including defending against any challenges in the United States Patent Office or other foreign patent offices to the scope, validity, and enforceability of such Patent or Patents, then Licensor shall no longer be responsible for the prosecution or maintenance of such Patent or Patents, or for defending against any challenges in the United States Patent Office or other foreign patent offices to the scope, validity, and enforceability of such Patent or Patents. If Licensee elects to assume responsibility for the prosecution or maintenance of such Patents, including defending against any challenges in the United States Patent Office or other foreign patent offices to the scope, validity, and enforceability of such Patents, any out-of-pocket amounts expended by Licensee therefor (including patent attorney's fees) shall be deemed to reduce Licensee's Net Revenues by an equal amount, for purposes of Licensee's Earned Royalties payment obligations.

9. Termination.

Licensor shall have the right to immediately terminate this Agreement if Licensee 9.1 shall at any time be in default in the payment of Earned Royalties hereunder or Licensee shall commit any other material breach of this Agreement and shall fail to remedy such default or material breach within thirty (30) days after receipt of written notice thereof by Licensor. In the event that Licensee, (i) admits in writing its inability to, or be generally unable to, pay its debts as such debts become due, (ii) applies for or consents to the appointment of, or the taking of possession by, a receiver, custodian, trustee, examiner or liquidator of itself or of all or a substantial part of its property or assets, (iii) makes a general assignment for the benefit of its creditors, (iv) commences a voluntary case under any applicable bankruptcy laws, (v) files a petition seeking to take advantage of any other law relating to bankruptcy, insolvency, reorganization, liquidation, dissolution, arrangement or winding-up, or composition or readjustment of debts, (vi) fails to controvert in a timely and appropriate manner, or acquiesces in writing to, any petition filed against it in an involuntary case under any applicable bankruptcy laws or (vii) takes any corporate action for the purpose of effecting any of the foregoing; then the Licensor may, by giving notice thereof to Licensee, terminate this Agreement as of the date specified in such termination notice. In the event of a termination of this Agreement, all rights with respect to the Patents shall automatically revert to Licensor. Termination of this Agreement shall be without prejudice to any moneys already paid or then due or to become due from Licensee to Licensor and without prejudice to any rights of either party accrued prior to the effective date of termination or that this Agreement indicates shall survive termination.

PATENT LICENSE AGREEMENT-Page 6

- 9.2 The provisions of Sections 8, 12 and 13 will survive the termination of this Agreement for any reason.
- 10. Injunctive Relief. In the event that Licensee materially breaches this Agreement, the parties agree that monetary damages may not provide an adequate remedy for Licensor. Therefore, in the event of a material breach or threat of material breach, Licensor shall be entitled to seek, in addition to any other relief available at law or in equity, injunctive relief to restrain Licensee from committing or continuing any violation of this Agreement, without the need for posting any bond or any other undertaking.

11. Disclaimer of Warranties.

Except as expressly set forth in this Agreement, nothing in this Agreement shall be construed as:

- 11.1 a warranty or representation by Licensor as to the validity, enforceability, or scope of any of the Patents;
- 11.2 a warranty or representation by Licensor that anything provided, made, used, imported, sold, or otherwise disposed under the license granted under this Agreement will not infringe patents, copyrights, trademarks, or other proprietary rights of third parties;
- 11.3 an obligation on Licensor to bring or prosecute actions or suits against third parties for infringement of the Patents;
- 11.4 a requirement that Licensor shall file or prosecute any patent application, secure any patent or maintain any patent in force and effect;
- 11.5 an obligation on the part of Licensor to furnish any manufacturing or technical information, or any information concerning the Patents;
- 11.6 grant by implication, estoppel, or otherwise any licenses, warranties (implied in fact or law) or rights other than those expressly granted herein, or creating any obligation other than those expressly created hereunder;
- 11.7 a representation, extension of any warranty of any kind, either express or implied or agreement by Licensor to assume any responsibility whatsoever with respect to use, sale, installation or disposition by the other parties or their vendees or transferees of any products or services of Licensee incorporating or made by use of Patents and other intellectual property licensed under the Agreement;
- 11.8 an obligation of Licensor to license any other patents or any other intellectual property rights not specifically licensed hereunder; or
- 11.9 grant of a right to use in advertising, publicity, or otherwise any trademark or trade name of another party.

PATENT LICENSE AGREEMENT-Page 7

- 12. Limitation of Remedies. NEITHER PARTY WILL BE LIABLE TO THE OTHER PARTY, OR ANY THIRD PARTY, FOR LOSS OF PROFITS, LOSS OF USE OR FOR ANY INDIRECT, SPECIAL, CONSEQUENTIAL, INCIDENTAL OR PUNITIVE DAMAGES WHETHER BASED UPON A CLAIM OR ACTION OF CONTRACT, WARRANTY, NEGLIGENCE, STRICT LIABILITY OR OTHER TORT, EVEN IF IT IS AWARE OF THE POSSIBILITY THEREOF. EACH PARTY AGREES THAT THE ENTIRE LIABILITY OF THE OTHER PARTY ARISING IN CONNECTION WITH THIS AGREEMENT WILL IN NO EVENT EXCEED AN AMOUNT EQUAL TO THE EARNED ROYALTIES PAID BY LICENSEE HEREUNDER.
- 13. Indemnification. Licensee shall defend, indemnify, and hold harmless Licensor, its affiliates and the directors, officers, employees, agents, successors and assigns of each of them, from and against all liability, obligations, costs, expenses (including reasonably attorneys' fees), losses, damages, penalties, judgments, claims, demands, actions, and suits, brought by or through any third party (hereinafter collectively called "liabilities") imposed on, incurred by or asserted against any one or more of them in connection with Licensee's use of the Patents or use, import, or sale of Licensed Equipment or Software, including but not limited to any liabilities arising out of actions in the form of tort, warranty, or strict liability, for death, personal injury, illness, or property damage. Licensee shall not settle or compromise any such claim, demand, lawsuit or other action in a manner that imposes any restrictions or obligations on Licensor or grants any rights to the Patents without Licensor's prior written consent.

14. Force Majeure.

- 14.1 In case of any failure to perform any obligations hereunder to the extent such failure is caused by war, acts of public enemies, fires, floods, governmental restrictions or actions, changes in the relevant laws and regulations, or any other force majeure, defaulting party shall immediately notify the other party thereof by written notification, and the notifying party will have no obligation to the other party.
- 15. Notices. Any notice or payment required to be given to either party must be sent to the respective addresses given below and is effective: (a) on the date of delivery if delivered in person, (b) five (5) days after mailing if mailed by first-class certified mail, postage paid, or (c) on the next business day if sent by overnight delivery. Either party may change its designated address by written notice given in accordance with this Section:

For Licensor:

CORE LABORATORIES GLOBAL NV

Attention: General Counsel 6316 Windfern Houston, Texas 77040 Facsimile: (713) 328-2152

PATENT LICENSE AGREEMENT-Page 8

Copy to:

Sherwood O. Jones Andrews Kurth LLP 600 Travis, Suite 4200 Houston, TX 77002 Fax: (713) 238-7113

For Licensee:

PARADIGM GEOTECHNOLOGY BV

Attention: General Counsel Chobham House Christchurch Way Woking Surrey GU21 1JP United Kingdom Fax: 44 (1483) 758001

Copy to:

John C. Ale Skadden, Arps, Slate, Meagher & Flom LLP 1600 Smith Street, Suite 4400 Houston, TX 77002 Fax: (713) 655-5200

- 16. Entire Agreement. This Agreement represents the entire understanding and agreement between the parties hereto relating to the subject matter hereof, and supersedes all prior and contemporaneous negotiations, agreements, and understandings with respect thereto. This Agreement may be modified or waived only by a separate writing signed by both parties expressly so modifying or waiving this Agreement.
- 17. Severability. If any provision hereof should be held invalid, illegal or unenforceable in any respect in any jurisdiction, then, to the fullest extent permitted by law: (a) all other provisions hereof shall remain in full force and effect in such jurisdiction and shall be liberally construed in order to carry out the intentions of the parties hereto as nearly as is possible; and (b) such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of such provision in any other jurisdiction. To the extent permitted by applicable law, Licensor and Licensee hereby waive their right to enforce any provision of law that would render any provision hereof prohibited or unenforceable in any respect. If the terms and conditions of this Agreement are materially altered as a result of this Section, the parties shall attempt to renegotiate the terms and conditions of this Agreement, in good faith, to resolve any inequities.

PATENT LICENSE AGREEMENT-Page 9

- 18. Governing Law. This Agreement shall be deemed to be executed and to be performed in the State of Texas, and shall be construed under and in accordance with the laws of the State of Texas, including but not limited to matters of validity, construction, effect, and performance. In the event of a controversy, claim, or dispute between the parties hereto arising out of or relating to this Agreement, such controversy, claim, or dispute may be tried exclusively in the courts of the State of Texas or in the United States Federal District Court located in, or closest to, Houston, Harris County, as either party may elect. Each of the parties hereby waives any defense of lack of in personam jurisdiction, improper venue, and forum non conveniens, and agrees that service of process of such court may be made upon each of them by personal delivery or by mailing certified or registered mail, return receipt requested, to the other party at the address provided for in Section 15 hereof. Both parties hereby submit to the jurisdiction of the court so selected, to the exclusion of any other courts which may have had jurisdiction apart from this Section 18.
- 19. Counterparts. This Agreement may be executed in one or more counterparts (including facsimile copies), each of which, when so executed, constitutes one original and all of which, when taken together, constitute one and the same Agreement.
- 20. Public Disclosures. Except that during the course of formal discovery or trial proceedings in any litigation or as otherwise legally required, no other public announcement or other disclosure to third parties concerning the existence or terms of this Agreement shall be made, either directly or indirectly, by either party hereto, without first obtaining the written approval of the other party, which shall include agreement upon the nature and text of such announcement or disclosure. The party desiring to make any such public announcement or other disclosure shall inform the other party of the proposed announcement or disclosure by providing the other party with a written copy thereof, and allowing reasonably sufficient time prior to public release to permit such other party to comment upon such announcement or disclosure. Once any such public announcement or disclosure has been made in accordance with this Section, then either party may appropriately communicate information contained in such permitted announcement or disclosure. Each party agrees that it shall reasonably cooperate with the other with respect to all disclosures regarding this Agreement to any other governmental or regulatory agencies, including requests for confidential treatment of proprietary information of either party included in any such disclosure, but neither party shall be prevented by this Section 20 from making any disclosures to such agencies required under applicable law. Notwithstanding anything to the contrary, this Agreement may be recorded (after redaction of its financial terms) in any national patent office in which a Patent is pending or has been issued or granted.
- 21. No Waiver. No failure on the part of a party to exercise, and no delay in exercising, any right shall operate as a waiver thereof, nor shall any single or partial exercise by a party of any right preclude any other future exercise thereof or the exercise of any other right.
- 22. Independent Contractors. No agency, partnership or joint venture is hereby established; each party shall act hereunder as an independent contractor. Neither Licensor nor Licensee shall enter into, or incur, or hold itself out to third parties as having authority to enter

into or incur on behalf of the other party any contractual obligations, expenses or liabilities whatsoever.

23. Transferability. Either party may transfer this Agreement to an Affiliate, or to an entity acquiring substantially all of the transferring party's assets, business or equity pertaining to the subject matter of this Agreement, upon written notification to the other party. Other than the foregoing, neither party may assign any of its rights, delegate any of its obligations, or transfer this Agreement, to any third party without written consent of the other party, and any such attempted assignment, delegation or transfer shall be null and void.

[Signature page follows.]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed as of the date first written above by their duly authorized representatives.

LICENS	OR:
	ABORATORIES GLOBAL NV, lands Antilles Company
Ву:	
Name:	
Title:	
LICENS	
LICENS	DEE:
PARAD	IGM GEOTECHNOLOGY BV
By:	
Pres	ident

EXHIBIT 1

LIST OF PATENTS

File No.	U.S. Serial/ Patent No.	Filing/ Issue Date	Title	STATUS/COMMENTS
COR-1021US	08/353,934 5,563,949	12/12/94 10/08/96	Method of Seismic Signal Processing and Exploration Inventors: Michael S. Bahorich; Steven L. Farmer	7.5 yr. Maintenance Fee: 4/8/04 11.5 yr. Maintenance Fee: 4/8/08 Expires 12/12/2014 Acquired from B.P. Amoco
COR-1021AG	334,022 250,730	10/27/95 6/18/97	Method of Seismic Signal Processing and Exploration Inventors: Michael S. Bahorich; Steven L. Farmer	Argentina Annuity Due: 06/18/04 Acquired from B.P. Amoco
COR-1021AU	41333/96 696,742	10/5/95 1/7/99	Method of Seismic Signal Processing and Exploration Inventors: Michael S. Bahorich; Steven L. Farmer	Australia Annuity Due: 10/5/04 Acquired from B.P. Amoco
COR-1021CA	2,179,901	10/5/95 8/18/98	Method of Seismic Signal Processing and Exploration Inventors: Michael S. Bahorich; Steven L. Farmer	Canada Annuity Due: 10/5/04 Acquired from B.P. Amoco
COR-1021CN	94191202.X 123,461	10/5/95 9/17/03	Method of Seismic Signal Processing and Exploration Inventors: Michael S. Bahorich; Steven L. Farmer	China Annuity Due: 10/5/04 Expiration: 10/5/15 Acquired from B.P. Amoco
COR-1021EG	869/95 20609	10/18/95 9/30/99	Method of Seismic Signal Processing and Exploration	Egypt Annuity Due: 10/21/04

PATENT LICENSE AGREEMENT- Page I- 1

File No.	U.S. Serial/ Patent No.	Filing/ Issue Date	Title	STATUS/COMMENTS
			Inventors: Michael S. Bahorich; Steven L. Farmer	Acquired from B.P. Amoco
COR-1021EP	95939565.8 0 736 185	10/5/95 7/9/97	Method of Seismic Signal Processing and Exploration Inventors: Michael S. Bahorich; Steven L. Farmer	European Patent Granted in France, Italy and the UK Annuity Due: 10/5/04 Acquired from B.P. Amoco
COR-1021MX	9603026 204,975	10/5/95 10/29/01	Method of Seismic Signal Processing and Exploration	12th-16th Year Annuity Due: 10/5/2005
			Inventors: Michael S. Bahorich; Steven L. Farmer	Expiration: 10/5/2015 Acquired from B.P. Amoco
COR-1021NO	1996.2731 311,316	10/5/95 11/12/01	Method of Seismic Signal Processing and Exploration	Norway Annuity Due: 10/5/04 Working Due: 11/12/2004
			Inventors: Michael S. Bahorich; Steven L. Farmer	Expiration: 10/5/2015
				Acquired from B.P. Amoco
COR-1021RU	96115275 2,144,683	10/5/95 1/20/00	Method of Seismic Signal Processing and Exploration	Russia Annuity Due: 10/5/04
			Inventors: Michael S. Bahorich; Steven L. Farmer	Acquired from B.P. Amoco
COR-1024US	08/716,612 5,838,564	10/01/96 1/17/98	Apparatus for Seismic Signal Processing and Exploration	7.5 yr. Maintenance Fee: 05/17/06 11.5 yr. Maintenance Fee: 05/17/10
				Expiration: 12/12/2014
				Continuation of 08/353,934, filed 12/12/94
		1		Acquired from B.P.

File No.	U.S. Serial/ Patent No.	Filing/ Issue Date	Title	STATUS/COMMENTS
***************************************				Amoco
COR-1022US	08/707,674 5,930,730 Now Reissue RE38,229	9/13/96 07/27/99 Reissued 08/19/03	Method and Apparatus for Seismic Signal Processing and Exploration Inventors: Kurt J. Marfurt; R. Lynn Kirlin; Steven L. Farmer; Michael S. Bahorich	7.5 yr. Maintenance Fee: 1/7/07 11.5 Maintenance Fee: 1/7/11 The "Semblance" patent Expiration: 12/12/2014 CIP of 08/353,934 filed 12/12/94 Acquired from B.P.
		1		Amoco
COR-1022AG	P960104613	10/4/96	Method and Apparatus for Seismic Signal Processing and Exploration Inventors: Kurt J. Marfurt; R. Lynn Kirlin; Steven L. Farmer; Michael S. Bahorich	Argentina: (in prosecution, office action response sent to AG patent office) Annuity Due: 10/5/04
				Acquired from B.P. Amoco
COR-1022AU	73829/96 709,621	9/30/96 12/16/99	Method and Apparatus for Seismic Signal Processing and Exploration	Australia: Annuity Due: 9/30/04
			Inventors: Kurt J. Marfurt; R. Lynn Kirlin; Steven L. Farmer; Michael S. Bahorich	Expiration: 9/30/2016
				Acquired from B.P. Amoco
COR-1022CA	2,204,168 Pending	9/30/96	Method and Apparatus for Seismic Signal Processing and Exploration	Canada: Annuity Due: 9/30/04
			Inventors: Kurt J. Marfurt; R. Lynn Kirlin; Steven L. Farmer; Michael S. Bahorich	Acquired from B.P. Amoco
COR-1022CN	96191172.7	9/30/96	Method and Apparatus for Seismic Signal Processing and Exploration	China: (First patent to issue, a divisional is in prosecution)
			Inventors: Kurt J. Marfurt; R. Lynn Kirlin; Steven L. Farmer; Michael S. Bahorich	

File No.	U.S. Serial/ Patent No.	Filing/ Issue Date	Title	STATUS/COMMENTS
				Acquired from B.P. Amoco
COR-1022EG	901/96 21656	10/5/96 1/30/02	Method and Apparatus for Seismic Signal Processing and Exploration Inventors: Kurt J. Marfurt; R. Lynn Kirlin; Steven L. Farmer; Michael S. Bahorich	Egypt Granted Annuity Due: 10/5/04 First Working: 1/30/05
				Acquired from B.P. Amoco
COR-1022EP	95936094.0 0796442 B1	9/30/96 12/12/01	Method and Apparatus for Seismic Signal Processing and Exploration Inventors: Kurt J. Marfurt; R. Lynn Kirlin; Steven L. Farmer; Michael S. Bahorich	European Patent for France, Italy, Netherlands and the UK Granted in: FR, IT, NL, UK Annuity Due: 9/30/04
				Acquired from B.P. Amoco
COR-1022MX	974,213 199,221	9/30/96 10/23/00	Method and Apparatus for Seismic Signal Processing and Exploration	Granted 10-14 Year Annuities Due: 9/30/05
			Inventors: Kurt J. Marfurt; R. Lynn Kirlin; Steven L. Farmer; Michael S. Bahorich	Expiration: 9/30/2016
			·	Acquired from B.P. Amoco
COR-1022NO	1997-1801 - Pending	9/30/96	Method and Apparatus for Seismic Signal Processing and Exploration Inventors: Kurt J. Marfurt; R. Lynn Kirlin; Steven L. Farmer; Michael S. Bahorich	Norway: (in prosecution, office action responded to by Schweppe in November 2003) Annuity Due: 9/30/04
				Acquired from B.P. Amoco
COR-1022RU	97108599 2,187,130	9/30/96 8/10/02	Method and Apparatus for Seismic Signal Processing and Exploration	Russia: Annuity Due: 9/30/04

File No.	U.S. Serial/ Patent No.	Filing/ Issue Date	Title	STATUS/COMMENTS
			Inventors: Kurt J. Marfurt; R. Lynn Kirlin; Steven L. Farmer; Michael S. Bahorich	Expires: 9/30/2016 Acquired from B.P. Amoco
COR-1025US	09/010,738 6,092,026	01/22/98 07/18/00	Seismic Signal Processing and Exploration	7.5 yr. Maintenance Fee: 01/18/08 11.5 yr. Maintenance Fee: 01/18/12
				Expiration: 12/12/2014
			Inventors Michael C. Rabariah	Continuation of 08/353,934, filed 12/12/94
			Inventors: Michael S. Bahorich; Steven L. Farmer	Acquired from B.P. Amoco
COR-1023US	08/815,496 5,892,732	03/11/97 04/06/99	Method and Apparatus for Seismic Signal Processing and Exploration	7.5 yr. Maintenance Fee: 10/6/06 11.5 yr. Maintenance Fee: 10/6/10
				Expiration: 4/12/2016
			L	CIP of 08/631,788, filed 4/12/96 (now abandoned)
			Inventor: Adam Gersztenkorn	Acquired from B.P. Amoco
COR-1023AG	P970101134	3/21/97	Method and Apparatus for Seismic	Annuity Due: 3/21/05
	AR 00634 B1	12/4/02	Signal Processing and Exploration	Expiration: 3/21/2017
			Inventor: Adam Gersztenkorn	Acquired from B.P. Amoco
COR-1023AU	18247/97	01/02/97	Method and Apparatus for Seismic Signal Processing and Exploration	
	710,968	01/20/00	Inventor: Adam Gersztenkorn	Expiration: 1/2/2017
			mventor. Adam Gersztenkom	Acquired from B.P. Amoco
COR-1023CA	2,220,274 Pending	01/02/97	Method and Apparatus for Seismic Signal Processing and Exploration	
			Inventor: Adam Gersztenkorn	responded to on 3/17/03

U.S. Serial/ Patent No.	Filing/ Issue Date	Title	STATUS/COMMENTS
			Acquired from B.P. Amoco
97190349.2 Pending	01/02/97	Method and Apparatus for Seismic Signal Processing and Exploration Inventor: Adam Gersztenkorn	In prosecution, office action responded to February 2003; annuities to run after grant of patent Acquired from B.P. Amoco
288/1997 21806	04/09/97 03/31/02	Method and Apparatus for Seismic Signal Processing and Exploration Inventor: Adam Gersztenkorn	Annuity due: 4/9/04 First Working: 3/31/05 Expires: 4/19/2017 Acquired from B.P.
97903759.5 Pending	01/02/97	Method and Apparatus for Seismic Signal Processing and Exploration Inventor: Adam Gersztenkorn	Amoco Annuity due: 1/2/05 In prosecution; office action sent September 2002 Acquired from B.P.
1997.5784 Pending	01/02/97	Method and Apparatus for Seismic Signal Processing and Exploration Inventor: Adam Gersztenkorn	Amoco Annuity due: 1/2/05 In prosecution; office action response sent August 2002 Acquired from B.P. Amoco
9710022 Pending	01/02/97	Method and Apparatus for Seismic Signal Processing and Exploration Inventor: Adam Gersztenkorn	In prosecution; annuities to run after grant of patent
97119642 2,169,931	01/02/97 06/27/01	Method and Apparatus for Seismic Signal Processing and Exploration Inventor: Adam Gersztenkorn	Acquired from B.P. Amoco Annuity due: 1/2/05 Expiration: 1/2/2017 Acquired from B.P. Amoco
			01/02/97 Method and Apparatus for Seismic Signal Processing and Exploration

File No.	U.S. Serial/ Patent No.	Filing/ Issue Date	Title	STATUS/COMMENTS
COR-1023T/T	970,138	11/11/97	Signal Processing and Exploration Inventor: Adam Gersztenkorn	Expiration: 1/2/2017 Acquired from B.P. Amoco

ANNEX 4

SUBLEASE AGREEMENT

[SEE ATTACHED SUBLEASE AGREEMENT]

EXHIBIT 4 TO PURCHASE AGREEMENT

SUBLEASE AGREEMENT

THIS SUBLEASE made as of the 22nd day of April, 2004.

BETWEEN:

CORE LABORATORIES CANADA LIMITED., a corporation having an office and carrying on business in the City of Calgary, in the Province of Alberta (hereinafter called the "Sublandlord")

OF THE FIRST PART

- and -

PARADIGM GEOPHYSICAL CANADA LTD., a corporation having an office and carrying on business in the City of Calgary, in the Province of Alberta (hereinafter called the "Subtenant")

OF THE SECOND PART

RECITALS

- A. By a Lease dated April 4, 2003 (a copy of which is attached hereto as Schedule "B") made between BTC Properties II Ltd., as landlord, and the Sublandlord, as tenant (the "Head Lease"), BTC Properties II Ltd. leased to the Sublandlord certain premises comprising a rentable area of approximately 30,782 square feet (the "Premises") located on the 17th and 21st floors of that building located at 125 9th Avenue SE, Calgary, Alberta (the "Building"), all as more particularly described in the Head Lease;
- B. Palliser Square Inc. (the "Head Landlord") is the successor in interest to BTC Properties II Ltd.;
- C. The Sublandlord has agreed to sublease and demise to the Subtenant a portion of the Premises consisting of approximately 23,089 square feet of rentable area to be located on all of the 17 floor and one-half of the 21st floor of the Building, as more particularly outlined in red on the attached Schedule "A" (the "Subleased Premises").

NOW THEREFORE, THIS SUBLEASE WITNESSES that in consideration of these premises and other good and valuable consideration, and the mutual covenants hereinafter contained, the Sublandlord and the Subtenant covenant and agree with each other as follows:

1. <u>DEFINITIONS</u>

Capitalized terms used herein and not otherwise defined shall have the meaning ascribed to such terms in the Head Lease.

474266.v3

2. <u>DEMISE AND SUBLEASE</u>

The Sublandlord does hereby demise and sublease to the Subtenant and the Subleassee does hereby accept the demise and sublease of the Subleased Premises, subject to the rents, covenants, agreements and conditions set forth herein. The Sublandlord and Subtenant shall agree upon (i) the distribution of the space in the Premises to be set forth on Schedule "A", which shall be initialed by both and attached hereto, and (ii) allocation of the telecommunications network situated in the Premises.

3. SUBLEASE TERM

The term of this Sublease shall be for a term of Nine (9) Years, Five (5) Months, Eight (8) Days commencing April 22, 2004 (the "Sublease Commencement Date") and expiring on September 29, 2013 (the "Sublease Term"). The Sublandlord and Subtenant expressly agree that the Sublease Term shall expire on September 29, 2013 and the Subtenant shall have no further right to occupy the Subleased Premises, nor shall the Subtenant be entitled to any right to extend the Sublease Term of this Sublease or expand the size of the Subleased Premises without the prior written consent and approval of the Sublandlord.

4. AREA

The Sublandlord and Subtenant agree that the rentable area of the Subleased Premises is 23,089 square feet, more or less, ("Rentable Area") which Subleased Premises are to be measured in accordance with the Head Lease prior to the Sublease Commencement Date or within a reasonable period of time thereafter.

5. ANNUAL RENT

The Subtenant hereby covenants and agrees to pay to the Sublandlord, or as the Sublandlord shall otherwise direct, in lawful money of Canada, net annual rent for the Subleased Premises ("Annual Rent") for and throughout the Sublease Term as follows:

- (a) for the period from the Sublease Commencement Date to September 30, 2005, Annual Rent shall be based upon a rate of Twelve Dollars (\$12.00) per annum per square foot of Rentable Area;
- (b) for the period from October 1, 2005 to September 30, 2008, Annual Rent shall be based upon a rate of Twelve Dollars and Fifty Cents (\$12.50) per annum per square foot of Rentable Area;
- (c) for the period from October 1, 2008 to September 30, 2010, Annual Rent shall be based upon a rate of Fifteen Dollars (\$15.00) per annum per square foot of Rentable Area; and

474266.v3

(d) for the period from October 1, 2010 to September 29, 2013, Annual Rent shall be based upon a rate of Sixteen Dollars (\$16.00) per annum per square foot of Rentable Area;

<u>provided</u>, however, that in the event that the Head Landlord grants any rebate, discount or concession with respect to Sublandlord's obligations to pay rent under the Head Lease, Subtenant's obligations to pay Annual Rent shall be reduced proportionately.

Annual Rent shall be payable in advance in equal monthly installments, without notice, and prorated for any partial months.

6. OPERATING COSTS AND REALTY TAXES

In addition to the Annual Rent, the Subtenant covenants and agrees that during the Sublease Term it will pay to the Sublandlord, or as the Sublandlord shall otherwise direct, its proportionate share of Operating Costs and Realty Taxes assessed against the Premises (as set forth in Section 2.07 of the Head Lease, attached hereto as Schedule "B") calculated for the Subleased Premises in accordance with the provisions of the Head Lease, as amended hereby. For greater certainty, the Subtenant covenants and agrees that the Subtenant's proportionate share of Operating Costs and Realty Taxes shall be calculated with reference to that fraction in which the total rentable area of the Subleased Premises is the numerator; and the total rentable area of the Premises is the denominator. Payments of Operating Costs and Realty Taxes shall commence on the Sublease Commencement Date.

Annual Rent and the Subtenant's proportionate share of Operating Costs and Realty Taxes are collectively referred to herein as "Rent".

7. TAXES

The Rent to be paid by the Subtenant to the Sublandlord is exclusive of any goods and services tax (herein called "G.S.T.") or other applicable value added tax as provided for in the Excise Tax Act, R.S.C. 1985, c. E-15, or in other similar legislation, exigible on the Rent, and the Subtenant shall pay to the Sublandlord concurrently with and in addition to the Rent payable hereunder an amount equal to the G.S.T. and such other tax applicable thereon.

8. APPLICATION OF HEAD LEASE

The Subtenant confirms that it has reviewed the Head Lease and covenants and agrees that to the extent amended herein, all of the terms and conditions contained in the Head Lease (save for the length of the term set forth therein, the amount of rent, the right to parking, the right of first refusal, the improvement allowance and the option to renew), shall apply to the use and occupation of the Subleased Premises, *mutatis mutandis* with reference therein to:

474266.v3

- (a) Landlord to be deemed to mean Sublandlord;
- (b) Tenant to be deemed to mean Subtenant;
- (c) Demised Premises to be deemed to mean the Subleased Premises;
- (d) Lease to be deemed to mean this Sublease;
- (e) Term to be deemed to mean the Sublease Term;

and, along with all of the paragraphs set out in this Sublease, shall be the complete terms and conditions of this Sublease.

9. SUBTENANT'S COVENANTS

The Subtenant covenants and agrees with the Sublandlord as follows:

- (a) it will pay to the Sublandlord the Rent due hereunder without any deduction, setoff or abatement;
- (b) except as same have been replaced by the provisions of this Sublease or amended in the attached copy of the Head Lease, it will perform all of the terms, covenants and conditions on the part of the Sublandlord to be observed and performed under the provisions of the Head Lease as they relate to the Subleased Premises;
- (c) it will use and occupy the Subleased Premises solely as general business offices; and
- (d) it will indemnify and save harmless the Sublandlord from and against all actions, expenses, claims and demands whatsoever arising out of any breach, default, or failure of the Subtenant to perform its covenants, obligations and agreements herein contained or arising out of its own negligence or wilful misconduct or the negligence or wilful misconduct of its employees, agents and independent contractors; provided that the Subtenant shall not be responsible for incidental or consequential or punitive damages, including loss of profits.

10. SUBLANDLORD'S COVENANTS

14

The Sublandlord covenants and agrees with the Subtenant as follows:

(a) upon payment by the Subtenant of the Rent, and other charges payable by the Subtenant pursuant to this Sublease and the performance, discharge and observance by the Subtenant of its covenants and agreements contained herein, the Subtenant shall be entitled to possess and enjoy the Subleased Premises

474266.v3

- peaceably for the Term, without any interruption or disturbance from the Sublandlord or any other person or persons lawfully claiming by, from or under it;
- (b) that the Sublandlord will comply with, perform, discharge and observe all of the terms, agreements, obligations and covenants, as tenant, pursuant to the Head Lease, except to the extent where same are the responsibility and obligation of the Subtenant pursuant to the terms of this Sublease; and
- (c) that it will take such commercially reasonably steps to cause to be performed all of the terms, covenants and conditions on the part of the Head Landlord to be observed and performed under the provisions of the Head Lease.

11. PARKING

The Sublandlord will request that the Head Landlord allocate to the Subtenant for the Subtenant's use Twenty (20) unreserved parking stalls over the Sublease Term at a rate of One Hundred and Fifty Dollars (\$150.00) per stall per month from the Sublease Commencement Date to September 30, 2005 and thereafter at market rates for the remainder of the Sublease Term. In addition, the Sublandlord will request that the Head Landlord allocate to the Subtenant for the Subtenant's use Seven (7) additional stalls on a month to month basis at current market rates.

12. SUBTENANT'S INSURANCE

The Subtenant hereby agrees to carry public liability and property damage insurance and Subtenant's legal liability insurance arising out of the use or occupation of the Subleased Premises in accordance with the Head Lease and agrees to provide to the Sublandlord proof of such insurance coverage prior to the commencement of the Term.

13. <u>ASSIGNMENT AND SUBLETTING</u>

This Sublease shall not be assigned by the Subtenant, nor shall the Subtenant sublet or otherwise part with the possession of the Subleased Premises in any manner whatsoever, without the Sublandlord's prior written consent and in accordance with the terms of the Head Lease.

<u>474266.v3</u>

14. NET SUBLEASE

The Subtenant hereby acknowledges, confirms and agrees that it is the intention of the Subtenant and the Sublandlord that, except as expressly set out herein, this Sublease shall be a completely net lease for the Sublandlord and that, subject to the provisions contained in the Head Lease, the Sublandlord shall not be responsible during the Sublease Term for any obligations, loss, charges, expenses or outlays of any nature and kind whatsoever arising from or relating to the Subtenant's use and occupation of the Subleased Premises or the contents thereof and that the Subtenant shall pay all such charges, impositions and expenses of every kind and nature relating to the Subtenant's use and occupation of the Subleased Premises and covenants with the Sublandlord accordingly. The Subtenant further acknowledges and agrees that the Sublandlord shall have all rights and remedies against the Subtenant in respect of this Sublease as though the Sublandlord was the landlord named in the Head Lease and the Subtenant was the tenant named in the Head Lease, *mutatis mutandis*, subject to the matters which are expressly dealt with in this Sublease.

15. CONDITION OF SUBLEASED PREMISES

The Subleased Premises shall be accepted by Subtenant on an "as is, where is" basis with respect to all existing leasehold improvements contained therein. At any time after occupancy of the Subleased Premises by the Subtenant, the Subtenant, at its sole risk and expense, shall be responsible for any alterations to the Subleased Premises provided that no alteration shall be made and no work done within or to the Subleased Premises without the Subtenant first having submitted the plans, working drawings and specifications therefore to the Sublandlord and the Head Landlord and having obtained the prior written approval from the Sublandlord and Head Landlord. The Subtenant shall comply with all the terms and conditions of such approval.

16. **DEFAULT**

In addition to the rights and remedies of the Sublandlord at law or as contained in the Head Lease, the parties agree that:

- (a) if the Subtenant does not pay to the Sublandlord the amounts required to be paid pursuant to the provisions of this Sublease, then the Sublandlord may make all such payments on behalf of the Subtenant, who shall forthwith reimburse the Sublandlord for all such payments, together with any costs and expenses incurred in connection therewith including interest as set forth in the Head Lease. In the event of any such non-payment by the Subtenant, the Sublandlord may take the same steps for the recovery of all such amounts as the Head Landlord may have for the recovery of Rent in arrears under the terms of the Head Lease, mutatis mutandis;
- (b) if the Subtenant is in default of any of its covenants hereunder (other than its covenant to pay Rent), the Sublandlord may take such steps and do or cause to be

474266.v3

done such things as may be necessary to remedy and correct the defaults and may thereupon charge its total costs and expenses incurred in respect thereof to the Subtenant who, hereby covenants and agrees to pay the same forthwith and the Subtenant hereby covenants that any and all such costs and expenses incurred by the Sublandlord and unpaid by the Subtenant shall be recoverable by the Sublandlord as if the same were, and in the same manner as, Rent reserved and in arrears under the terms of this Sublease; if the Sublandlord shall be entitled to, and shall elect to make a re-entry as provided for in the Head Lease, the re-entry or other action so taken shall not be deemed to relieve the Subtenant of its obligation to pay Rent and other monies payable as Rent or otherwise in accordance with the provisions hereof, all of which shall continue to accrue and be payable until such time as the Sublandlord is able to re-let the Subleased Premises or otherwise deal with the same in such manner that it would not sustain any loss should the Subtenant thereafter fail to pay the Rent and other monies payable as Rent or otherwise under this Sublease.

17. NOTICES

(a) Any notices or demands of any kind which are required to be given under this Sublease shall be in writing and shall be delivered to the address set out below for the party to whom it is delivered, in the case of the Sublandlord, addressed to it at:

CORE LABORATORIES CANADA LIMITED 2810 12th Street N.E. Calgary, Alberta, T2E 7P7

Attention: Controller

and in the case of the Subtenant, addressed to it at:

PARADIGM GEOPHYSICAL CANADA LTD.

at the Subleased Premises

or such other address as a party may from time to time advise the other party by notice in writing.

(b) The Rent and all other amounts to be paid by the Subtenant to the Sublandlord shall be paid to the Sublandlord at its address referred to in subsection (a) above, or to such other person or persons at such place or places as the Sublandlord may from time to time designate in writing.

474266.v3

18. GENERAL

- (a) Subject to the provisions of this Sublease respecting assignment, this Sublease shall enure to the benefit of and be binding upon the Sublandlord and the Subtenant and their respective successors and permitted assigns.
- (b) In the event that there is any conflict between the terms of the Head Lease and the terms and conditions contained in this Sublease, then the terms and conditions contained in this Sublease shall govern.
- (c) This Sublease shall be construed in accordance with and governed by the laws of the Province of Alberta.
- (d) This Sublease may be executed in counterparts and when each party has executed a counterpart, each of such counterparts shall be deemed to be an original and all of such counterparts when taken together shall constitute one and the same agreement.
- (e) Time will be of the essence in this Sublease.
- (f) This Sublease or a counterpart hereof may be executed and transmitted by fax, with transmission confirmed as complete, and if so executed and transmitted, this Sublease shall be for all purposes as effective and binding upon such party as if such party had delivered an originally executed document. A party transmitting an executed document by telecopy shall forthwith deliver the original of the executed document.

19. ENTIRE AGREEMENT

This Agreement contains the entire agreement between the parties in respect of this Sublease and there are no other warranties, representations, terms, conditions or collateral agreements, express or implied, between the parties in respect thereof.

20. SUBLANDLORD'S CONDITIONS PRECEDENT

This Sublease is conditional upon the Sublandlord obtaining the Head Landlord's consent to the sublease of the Subleased Premises by the Sublandlord to the Subtenant. In the event the aforesaid conditions precedent are not fulfilled, this agreement shall be null and void and the parties shall not owe any obligation to the other.

474266.v3

IN WITNESS WHEREOF the parties hereto have executed this Sublease as of the day first above written.

Sublandlord:

CORE CANADA I	LABORATORIES LIMITED
Per:	
Per:	
Subtenant:	
PARADIO	GM GEOPHYSICAL CANADA
LTD	
Per:	Name: Shai Bible
Per:	Tille: Attorner in feet
Per:	Tille: Attorne infect

474266.v3

SCHEDULE "A" SUBLEASED PREMISES

474266.v3

SCHEDULE "B"

HEAD LEASE

474266.13

PAGES 1-42

REDACTED

EXHIBIT 2.1(f)

INTELLECTUAL PROPERTY

- 1. See RTD Patent Portfolio Status Summary updated 3/25/04 attached.
- 2. See RTD Trademarks Status Summary updated 3/25/04 attached.
- 3. See Exhibit 2.1(h) Licenses for Licenses of Intellectual Property.
- 4. ADS Depth Software.

-43-

44-

٠	•	•	
•			
1	1		•
¢	۶	Ć	۲
4	•	۰	١
¢	٠	•	
ŧ		,	
1	۹	,	ì
ŧ	•	•	į
	٠		•
1	•	۰	٠
			•
		Ļ	

		· 첫 .							
STATUS/COMMENTS	3.5 yr. Maintenance Fee: 11/292004 7.5 yr. Maintenance Fee: 11/29/2008 11/29/2012 Expiration: 06/12/17	Prosecution by Winstead, Sechrest & Minick, P.C. Acquired from Coherence Technology Co., 1nc.	7.5 yr. Maintenance Fee: 10/25/07 11.5 yr. Maintenance Fee: 10/25/11	Expiration: 10/09/18	Prosecution by Browning Bushman Acquired from Coherence Technology Co., Inc.	3.5 yr. Maintenance Fee: 2/6/06 7.5 yr Maintenance Fee: 2/6/10	11.5 yr Maintenance Fee: 2/6/14 Expiration: 10/20/20	Annuity Due: 10/20/04 Request Examination: 10/20/05	and the second s
Title	Method and Apparatus for Data Channelization and Hardware-Based Network Operation and Control	Inventors: Robert C. Wesolek; Kevin F. Fotorny	Method of Selsmic Signal Process		Inventors: Vasudhaven Sudhakar, Robert S. Stephens	Transfer Function Method of Seismic Signal Processing and Exploration	Inventors: Vasudhaven Sudahakar; Emil Blias	Transfer Function Method of Seismic Signal Processing and Exploration	Inventors: Vasudhaven Sudahakar; Emil Blias
Filing/Issue Date	06/12/97 05/29/01		10/9/98 4/25/00			10/20/00 8/6/02		10/20/00	, programme and
U.S. Serial/ Patent No.	08/874,013 6,240,097		09/169,403 6,055.482			09/693,610 6,430,508		2,387,760 Pending	Andrew
File No.	COR-1083		COR-1035US			COR-1014US		COR-1014CA	
Owned/ Licensed	Owned		Owned			Owned		Owned	

RTD PATENT PORTFOLIO STATUS SUMMARY updated April 19, 2004

Œ	File No.	U.S. Serial/ Patent No.	Filing/Issue Date	Title	STATUS/COMMENTS
COR-1043US	શ્	10/324,164 Pending	12/19/02	Method for High Frequency Restoration of Seismic Data	Pending HFR method
				inventors: Vasudhaven Sudahakar, Satinder Chopra	
COR-1043PCT	звст	PCT/ US03/39988 Pending	12/19/03	Method for High Frequency Restoration of Seismic Data	Pending Demand due 07/19/04
				Inventors: Vasudhaven Sudahakar; Satinder Chopra	
COR-1044US	44US	60/412,486 10/667,037 Pending	09/20/02 09/22/03	Method of Reconstructing Seismic Records to Obtain High Resolution Signals	Pending
				Inventors: Jason Choo, Jonathon Downton, Janet Dewar	
JR-40	COR-1044PCT	PCT/ US03/29619 Pending	09/22/03	Method of Reconstructing Seismic Records to Obtain High Resolution Signals	Pending Demand due 04/22/04
			PORRE OF THE STATE	Inventors: Jason Choo, Jonathon Downton, Janet Dewar	Other is a dead and a dead and an experimental and a dead of the second
)R-1(COR-1044VE	2003-001616 Pending Venezuela	09/19/03	Method of Reconstructing Seismic Records to Obtain High Resolution Signals	Pending
		The second secon	Application of the state of the	Inventors: Jason Choo, Jonathon Downton, Janet Dewar	THE POST OF THE PO
)R-10	COR-1058US	60/416,140 10/679,923 Pending	10/04/02 10/06/03	Method and Apparatus for Seismic Data Processing System	Pending
		Amalana de	- And	Inventor: J. Bee Bednar	

Owned/ Licensed	File No.	U.S. Serial/ Patent No.	Filing/Issue Date	Title	STATUS/COMMENTS
Owned	COR-1058PCT	PCT/ US03/31835 Pending	10/06/03	Method and Apparatus for Seismic Data Processing System	Pending
				Inventor: J. Bee Bednar	
Owned	COR-1060US	60/416,068 10,679,890 Pendin	10/04/02 10/06/03	Method and System for Distributed Tomographic Velocity Analysis Using Dense P-Maps	Pending Missing parts due 4-2-04 Bournost for Doctoration and Assignment sont
		D		Inventors: J. Bee Bednar; Houzhu Zhang; Chengbin Peng	to Houzhe and Peng
Owned	COR-1060PCT	PCT/ US03/31478	10/06/03	Method and System for Distributed Tomographic Veiocity Analysis Using Dense P-Maps	Pending Demand due 05/04/04
:		renaing		Inventors: J. Bee Bednar; Houzhu Zhang; Chengbin Peng	
Owned	COR-1079US	10/646,195 Pending	08/22/03	Method to Extend Seismic Bandwidth and Decrease Limit of Resolution	Pending
				Inventors: Alvaro Chaveste; Jaime Stein	

47-

HOU:2293083.11

RTD TRADEMARKS STATUS SUMMARY updated April 19, 2004

Owned/ Licensed	File No.	Mark	Main Operating Entity	Filing/ Registration Date	Serial/ Registration No.	Next Action Due Date	Comments
Owned	COR-4034UK	G.W.I.S (Class 9)			1,366,830	12/14/05	10 Year Renewal Acquired from Scott Pickford
Оwпеd	COR-4040US	REBEL	Core Lab (ADS)	4/28/99 5/21/02	75/692,925 2,571,839	11/21/07 5/21/12	Section 8 and 15 10 Year Renewal Acquired from Advanced Data Solutions
Owned	COR-4047US	Z-RIG	Core Lab	3/18/03	78/226,802		In prosecution: Notice of Allowance has been issued
Owned	COR-4048US	INTERPRETER-GUIDED	Core Lab	3/19/03	78/227,263		In prosecution: Notice of Allowance has been issued
Owned	COR-4053	T-H-T	Core Lab	9/30/03	78/307,747	TATALIHAT TATALI	In prosecution
Owned	COR-4054	HFR	Core Lab	9/30/03	78/307,774		In prosecution
Owned	COR-4055	FBE	Core Lab	9/30/03	78/307,784	A DE LA REAL AND	In prosecution
Owned	COR-4056	LAMBDA RECONSTRUCTION	Core Lab	\$0/2/03	78/310,501		In prosecution

PAGES 48-68

REDACTED

PATENT REEL: 028590 FRAME: 0107

RECORDED: 07/19/2012